TRANSDIMENSIONAL PLASMONIC TITANIUM NITRIDE FOR TAILORABLE NANOPHOTONICS

by

Deesha Shah

A Dissertation

Submitted to the Faculty of Purdue University In Partial Fulfillment of the Requirements for the degree of

Doctor of Philosophy



School of Electrical and Computer Engineering West Lafayette, Indiana May 2022

THE PURDUE UNIVERSITY GRADUATE SCHOOL STATEMENT OF COMMITTEE APPROVAL

Dr. Alexandra Boltasseva, Chair

School of Electrical and Computer Engineering

Dr. Vladimir M. Shalaev

School of Electrical and Computer Engineering

Dr. Alexander Kildishev

School of Electrical and Computer Engineering

Dr. Zubin Jacob

School of Electrical and Computer Engineering

Approved by:

Dr. Dmitrios Peroulis

To Mom, Dad, and Akhil

ACKNOWLEDGMENTS

I would like to first thank my co-advisors, Professors Sasha Boltasseva and Vlad Shalaev, for their exceptional mentorship and constant encouragement. I greatly appreciate all the support and feedback that they provided during my Ph.D. Thank you to both for always inspiring us to reach our potential and continue to grow as researchers. I am very grateful for being a part of their academic family. I would also like to thank Professors Alexander Kildishev and Zubin Jacob for agreeing to be a part of my committee.

During my time at Purdue, there were many people who helped me along the way. I'd like to thank Professor Nate Kinsey for his useful guidance and for mentoring me during my initial experiments when I first joined the group. I would also like to thank my collaborators Dr. Arrigo Calzolari and Professor Igor Bondarev as well as Dr. Zhaxylyk Kudyshev for our fruitful discussions, which helped to strengthen my understanding of the underlying physics behind my experiments.

All of my group members were always extremely supportive during my time at Purdue. I express my sincere appreciation to all of my current and former colleagues including Dr. Alexei Lagoutchev, Prof. Simeon Bogdanov, Prof. Justus Ndukaife, Dr. Misha Shalaginov, Dr. Di Wang, Dr. Clayton Devault, Dr. Krishnakali Chaudhuri, Dr. Sajid Choudhury, Dr. Shaimaa Azzam, Dr. Zhuoxian Wang, Dr. Urcan Guler, Dr. Aveek Dutta, Dr. Harsha Reddy, Dr. Soham Saha, Sarah Nahar Chowdhury, Oksana Makarova, Sam Peana, Xiaohui Xu, Mustafa Goksu Ozlu, Zach Martin, Omer Yesilyurt, Morris Yang, Bruce Ding, Dr. Demid Sychev, Dr. Colton Fruhling, Dr. Alexander Senichev, Dr. Vahagn Mkhitaryan, Mira Marinova, Karthik Pagadala, Brandon Triplett, and Yuheng Chen.

I'm thankful to all my friends in Birck who made my experience at Purdue so joyous and fun. I especially thank my close friends Soham, Sarah, and Oksana for always being there for me and having my back throughout my Ph.D. I will forever cherish our many adventures together and be grateful for the lifelong friends I made in this research group.

I'd also like to express my gratitude to Sumanth for his understanding and patience throughout this entire time. Over these past few years, he has shown me tremendous love and support, which kept me going until the finish line. Finally, I would like to thank Mom, Dad, Akhil, and Sapna for their unconditional love and support every step of the way from the very beginning. I thank my parents for teaching me the value of perseverance and hard work, which helped me complete this journey. I would not have made it this far without my family constantly cheering me on.

TABLE OF CONTENTS

LI	ST O	F TAB	LES	9
LI	ST O	F FIGU	JRES	10
A	BSTR	ACT		14
1	TRA	NSDIN	IENSIONAL PLASMONIC MATERIALS	15
	1.1	Optica	al Properties of TDMs	16
	1.2	Strong	gly Confined Plasmons in TDMs	18
	1.3	Confin	ement Enhanced Optical Nonlinearities	19
	1.4	Tuning	g the Properties of Metals	21
	1.5	Mater	ial Platforms for Transdimensional Plasmonics	22
	1.6	Outlin	.e	24
2	ОРТ	TICAL 1	PROPERTIES OF ULTRA-THIN TITANIUM NITRIDE	26
	2.1	Introd	uction	26
	2.2	Metho	ds	27
		2.2.1	Material Growth	27
		2.2.2	Optical Characterization	28
		2.2.3	Electrical Characterization	30
	2.3	Result	S	32
		2.3.1	Optical Properties of Ultrathin TiN	32
		2.3.2	Scattering Losses in Ultrathin TiN	34

		2.3.3	Thickness Dependent Drude Plasma Frequency	35
		2.3.4	Thickness Dependent ε''	36
		2.3.5	Role of Effective Mass in the Thickness Dependent Optical Properties	37
		2.3.6	Thickness Dependent Resistivity	38
	2.4	Summ	ary	39
3	FIRS	ST PRI	NCIPLE STUDIES OF ULTRATHIN TITANIUM NITRIDE	41
	3.1	Introd	uction	41
	3.2	Metho	ds	41
		3.2.1	Density Functional Theory (DFT)	41
		3.2.2	Material Growth	43
		3.2.3	X-ray Photoelectron Spectroscopy	43
		3.2.4	Optical Characterization	44
	3.3	Result	s and Discussion	44
		3.3.1	DFT Modelling of the Electrical and Optical Properties of Ultrathin	
			TiN	44
		3.3.2	Effect of Surface Oxidation	49
		3.3.3	Effect of Substrate Induced Strain	53
		3.3.4	Effect of the MgO Substrate	56
	3.4	Summ	ary	58
4	THI	CKNES	S DEPENDENT PLASMONIC PROPERTIES OF PASSIVATED TRAN	S-
	DIM	ENSIO	NAL TITANIUM NITRIDE	60

	4.1	Introd	uction	60		
	4.2	Metho	ds	61		
		4.2.1	Material Growth	61		
		4.2.2	Optical Characterization	63		
	4.3	Result	s and Discussion	64		
		4.3.1	Thickness Dependent Complex Dielectric Permittivity	64		
		4.3.2	Drude Plasma Frequency	69		
		4.3.3	Thickness Effect on Drude Damping	73		
		4.3.4	Drude Damping in Unpassivated Films	76		
	4.4	Summ	ary	77		
5	TRA	NSDIN	IENSIONAL TITANIUM NITRIDE FOR ACTIVE METASURFACES	79		
	5.1	Introd	uction	79		
	5.2	Electri	ical Tunability of Atomically Thin TiN	80		
	5.3	Phase	Modulation with Atomically Thin TiN	81		
	5.4	Summ	ary	82		
6	Cone	clusion a	and Outlook	84		
RI	EFER	ENCES	\$	86		
VI	VITA 102					
ΡĮ	JBLIC	CATION	NS	103		

LIST OF TABLES

2.1	Ellipsometry fitting parameters for ultra-thin TiN using a Drude - Lorentz model	32
4.1	Drude - Lorentz fitting parameters for passivated TD TiN films $\ldots \ldots \ldots$	65

LIST OF FIGURES

2.1	AFM images of (a) 2 nm, (b) 6 nm, and (c) 8 nm TiN films showing a very low roughness of <0.21 nm, indicating atomically smooth films	28
2.2	Ellipsometry measurements, Ψ and Δ , of the ultrathin films. (a) and (c) are Ψ and Δ for 50°, while (b) and (d) are Ψ and Δ for 70°	29
2.3	Optical image of the an 8 nm Hall cross with the gold contact pads used for extracting the charge carrier densities. The Hall cross was fabricated using photolithography and a dry etching process.	31
2.4	Thickness dependent (a) real and (b) imaginary parts of the dielectric per- mittivity as retrieved from ellipsometry measurements. The films become less metallic with decreasing thickness.	33
2.5	Carrier densities obtained from Hall measurements and damping factor from ellipsometry measurements. As the thickness decreases, the metallic character decreases and the scattering losses increase. The error bar for the carrier densities is obtained from the error in the sputtering rate	34
2.6	Plasma frequency extracted from the ellipsometry fittings (blue) compared to the plasma frequency calculated assuming an average effective mass (red). Below 4 nm, the calculated plasma frequency is outside of the error bar for the ellipsometry values and begins to deviate from the experimental trend, which may be a result of either modifications in the band structure caused by the oxynitride or quantum confinement effects.	36
2.7	(a) Calculated effective mass using Hall measurements and ellipsometry re- sults. (b) Plasma frequency squared as a function of carrier density, which is observed to be linearly proportional, indicating that the effective mass may be relatively constant. The slope of (b) is used to calculate the average effective mass	38
2.8	Optical and electrical resistivity retrieved from ellipsometry and four-point probe measurements, respectively. Increases in the resistivity is due to the lower carrier concentration due to oxidation and a higher losses due to more surface scattering in thinner films as a result of Differences in the calculated resistivity is due to the different scattering effects that contribute to each technique	40
3.1	(a) Side view of 1-10, 15 and 20 layer thick $\text{TiN}(100)$ slabs used in the simulations. (b) free electron charge density $n_{\rm e}$, and (c) volume-plasmon E_p and screened-plasmon E_0 energies for 1-20 layer films. Although the 20 monolayer thick TiN is highly metallic, the bulk limit for the plasmon energies is not yet reached. The orange shaded area estimates the error bar associated with the choice of m^*	45

3.2	(a) Side view of 1-10, 15 and 20 layer thick $\text{TiN}(100)$ slabs used in the simulations. (b) free electron charge density $n_{\rm e}$, and (c) volume-plasmon E_p and screened-plasmon E_0 energies for 1-20 layer films. Although the 20 monolayer thick TiN is highly metallic, the bulk limit for the plasmon energies is not yet reached. The orange shaded area estimates the error bar associated with the choice of m^*	46
3.3	Screened plasmon energy E_0 as a function of the number of TiN(100) layers (N_L)	47
3.4	Simulated DOS plots for 1L (red) and 10L (black) systems. Dashed lines remark energy position with respect to the Fermi level (E_F) and the energy width of the two main groups of bands (1, 2) responsible for crossover wave- length in the visible range (λ_0) . Due to electron confinement, a shrinking in the energy bandwidth is observed, resulting in the red shift of the crossover wavelength observed in Figure 3.2	48
3.5	(a) XPS spectrum of a partially oxidized 30nm TiN film grown on MgO by DC magnetron sputtering. The (b) N_{1s} , (c) O_{1s} , and (d) Ti_{2p} peaks are fitted to determine the local chemical environment. CPS are the counts per second collected at each energy. The black curve represents the actual XPS data collected, while the blue dashed curve is the generated envelope curve from peak fitting. The films consist of three phases: TiN, TiN_xO_y , and TiO_2	51
3.6	(a) Configurations for single oxygen atom in a nitrogen substitutional site (ON) , single oxygen atom on the surface (O_s) or diatomic oxygen molecule on the film surface (O_2) . Real part of the dielectric function of oxidized TiN film including (b) single atoms ON and Os; and (c) O_2 molecule. Different combinations of these three main configurations are modeled to determine the effect of oxygen in the films. Theoretical ideal model (10L) and the corresponding experimental spectra are reported in panels b and c for direct comparison.	52
3.7	(a) Relaxed geometry of $3O_N + 2O_2$ configuration. (b) Total (blue line) and O- projected (shaded area) density of states of $3O_N + 2O_2$ configuration compared to the ideal unoxidized 10L case (black line). Zero energy reference is set to the Fermi level of the two systems	53
3.8	(a) In-plane strain model. (b) Simulated real part of the dielectric function for 10L TiN films, with varying amount of applied lateral strain. The crossover wavelength red shifts with tensile strain. Theoretical ideal unstrained model (10L) and corresponding experimental spectra are reported in panels (b) for direct comparison.	54
3.9	HAADF STEM image of a 2 nm TiN film on an MgO substrate. The contrast at the film/substrate interface is due to misfit dislocations, causing the film to relax. This indicated that we do not have any significant contributions from strain in the experimentally characterized optical properties	55

3.10	Real part of the dielectric function extracted from ellipsometry measurements for ultrathin TiN films grown on MgO and c-sapphire substrate	56
3.11	Simulated TiN/MgO interface: a) atomic structure (side view); b) real part of the dielectric function; c) total (black area) and TiN- (cyan area) and MgO-projected (red line) density of state. Vertical dashed line marks the Fermi Level (E_F) of the interface.	57
4.1	AFM images of a 10 nm AlScN film on (a) 2 nm thick TiN film and (b) 8 nm thick TiN film with a root mean square (RMS) roughness of 2.8 Å and 2.7 Å, respectively. The low roughness indicates uniformly passivated and atomically smooth films.	61
4.2	High-angle annular dark field imaging scanning transmission electron microscopy (HAADF STEM) image of an epitaxial 1 nm (5 monolayers) TiN film on MgO. The inset shows the diffraction pattern revealing the high degree of crystallinity of ultrathin TiN	62
4.3	Complex permittivity of the AlScN passivation layer used in the ellipsometry fittings for the passivated TiN films	63
4.4	Permittivity of the MgO substrate used in the ellipsometry fittings \ldots .	64
4.5	Complex dielectric permittivity for different thicknesses measured by ellip- sometry. (a) Real part of the permittivity indicates lower metallicity with decreasing thickness. (b) Imaginary part increases with smaller thickness, indicating higher losses in thinner films	66
4.6	Thickness dependent epsilon-near-zero (ENZ) wavelength, which redshifts in thinner films. The experimental trend is consistent with the findings from the DFT studies presented in Chapter 3	67
4.7	Comparison between the real part of the permittivity for unpassivated (dashed) and passivated (solid) films with thicknesses of 2 (black) and 4 nm (red). Passivated films have a larger $ \varepsilon' $, indicating a stronger metallic character compared to unpassivated films because of the absence of oxidation effects.	68
4.8	(a) The plasma frequency data points extracted from the ellipsometry spectra measured for ultrathin TiN films of different thicknesses. (b) The data point set of (a) normalized. The black dashed line shows the ratio $\bar{\omega}_p(d))/\omega_p^{3D}$ as given by Eqn. 4.5with parameter $C = 7.7 \text{ nm}^{-1}$. The theory agrees very well to the experiments, indicating that our experiments demonstrate confinement induced optical properties in TD TiN	73
4.9	Thickness dependent Drude damping factor from ellipsometry measurements (black squares), from calculations (blue dashes), and calculated Fermi ve- locity, v_f (red dashes). The increase in Drude damping in thinner films is a result of an increase in surface scattering and larger contribution from surface roughness effects.	74

4.10	AFM image of an unpassivated 4 nm TiN film showing atomically smooth surface with a roughness of 2.1 Å. The surface roughness of unpassivated films is very similar to passivated films, indicating that the contribution of surface scattering to the Drude damping should be similar.	76
4.11	Comparison between the thickness dependent Drude damping of passivated (red) and unpassivated (blue) ultrathin TiN films, as extracted from the el- lipsometry fittings. Due to similar surface roughness in both types of films, the trends in Drude damping are quantitatively very similar for passivated and unpassivated films.	77
5.1	Calculations of relative changes in carrier densities achievable in TiN through electrical gating for varying film thicknesses. Close to $15~\%$ relative change is expected in case of monolayer TiN \ldots	80
5.2	(a) Experimentally retrieved carrier density as a function of bias gate volt- age in 1-nm-thick TiN film with close to 8.5% change in carrier density; (b) schematic of the gate dependent Hall measurements. A 35-nm-thick layer of Al_2O_3 is used as the gate oxide and gold as the back-gate electrode	81
5.3	(a) Schematic of Au strip-TiN film hybrid metasurface. (b) Change in reflection phase and coefficient at 1.550 μ m as a function of the change in charge carrier density. A phase modulation of >300° is achieved by lower the charge carrier density from 6% lower to 9%	82

ABSTRACT

In the realm of tunable optical devices, 3D nanostructures with metals and dielectrics have been utilized in a wide variety of practical applications ranging from optical switching to beam-steering devices. 2D materials, on the other hand, have enabled the exploration of truly new physics unattainable with 3D systems due to quantum confinement leading to unique optical properties and enhanced light-matter interactions. Transdimensional materials (TDMs) – atomically thin films of metals – can couple the robustness of 3D nanostructures with the new physics enabled by 2D features. However, the evolution of the optical properties in the transdimensional regime between 3D and 2D is still underexplored. The optical properties of metallic TDMs are expected to show unprecedented tailorability, including strong dependences on the film thickness, composition, strain, and surface termination. They also have an increased sensitivity to external optical and electrical perturbations, owing to their extraordinary light-confinement. Additionally, the small atomic thicknesses may lead to strongly confined surface plasmons and quantum and nonlocal phenomena. The strong tunability and light-confinement offered by TDMs have resulted in a search for atomically thin plasmonic material platforms that facilitate active metasurfaces with novel functionalities in the visible and near infrared (NIR) range. In this research, we explore the plasmonic properties and tailorability of atomically thin titanium nitride (TiN). We experimentally and theoretically study the thickness-dependent optical properties of epitaxial TiN films with thicknesses down to 1 nm to demonstrate confinement induced optical properties. Overall, this research demonstrates the potential of TDMs for unlocking novel optical phenomena at visible and NIR wavelengths and realizing a new generation of atomically thin tunable nanophotonic devices.

1. TRANSDIMENSIONAL PLASMONIC MATERIALS

Surface plasmons (SPs), which are optically induced oscillations of the electron cloud excited at metal-dielectric interfaces, enable strong light confinement beyond the diffraction limit and highly localized electromagnetic fields[1], [2]. Plasmonic nanostructures uniquely allow for light manipulation at the nanoscale and have been instrumental in advancing a range of applications [3], including photovoltaics [4], [5], optical circuitry [6]–[8], and imaging [9], [10]. Advancements in nanofabrication techniques have allowed for precise optimization of the optical response of plasmonic devices by controlling the properties of the material building blocks, as well as the structure dimensions and geometry at the nanoscale [11]. Moreover, the dynamic control of the plasmon resonances has been of great interest to produce tunable nanophotonic devices for optical switches [12], [13], sensing [14]–[16], and modulators [17], [18].

Plasmonic components are typically made up of metals, such as gold or silver, due to their low losses and high conductivity [19]. However, because of their large charge densities, the penetration of electromagnetic fields inside metals is very low. This low penetration depth, coupled with the high ohmic losses, makes appreciable optical tuning difficult with bulk metals [20], [21]. Additionally, noble metals are known to be extremely soft with a low mechanical and thermal stability, limiting their applications for harsh environment devices. Noble metals' softness, lack of tunability, and low melting point, along with challenges in thin film fabrication, have prevented the realization of practical plasmonic devices requiring tailorable optical properties and durable metallic nanostructures. In response to these pressing requirements, investigations into new plasmonic materials that meet the need for tunable and robust nanophotonic devices have increased.

Emerging plasmonic materials, such as transparent conducting oxides (TCOs) [22]–[25] and graphene [17], [26], [27], have been heavily investigated for their highly tunable response. In the case of TCOs, a low carrier concentration compared to bulk metals[24], [28] enables a greater field-penetration into the materials with lower optical losses. This enhances the light-matter interaction, resulting in strong modulation. On the other hand, due to the monolayer thickness of two-dimensional (2D) materials, large changes in the sheet carrier density are

easily attained [29]. In addition to their tunability, the strong quantum confinement in 2D materials has led to unique phenomena unattainable with conventional bulk plasmonic materials, including forbidden atomic transitions, strong nonlinearities, and highly confined plasmons [30], [31].

Similar to 2D materials, ultra-thin films in the transdimensional (TD) regime spanning a few atomic layers, known as transdimensional materials (TDMs), are expected to exhibit strong dependencies on structural and compositional properties, as well as extreme sensitivities to external optical and electrical perturbations [32], [33]. Additionally, the small atomic thicknesses may lead to strongly confined surface plasmons and unique quantum phenomena like forbidden atomic transitions and large nonlinearities[30], [34]. While many of these unique effects have been demonstrated in 2D materials like graphene, the plasmon modes of these materials are at lower frequencies in the mid-infrared (MIR) range, far from the technologically important near-infrared (NIR) and visible wavelengths. Metallic TDMs are highly promising for achieving similar quantum and nonlocal effects in the visible and NIR. The strong tunability and light confinement offered by TDMs has resulted in a search for atomically thin material platforms that facilitate active metasurfaces with novel functionalities in the visible and NIR range [32]. TDMs are highly promising for greatly advancing plasmonic functionalities through unparalleled tailorability, dynamic control, and exceptionally high localization of light.

The content presented in this chapter is taken from the paper D. Shah, Z. A. Kudyshev, S. Saha, V. M. Shalaev, and A. Boltasseva, "Transdimensional material platforms for tunable metasurface design," MRS Bulletin, 45 (3), 2020, D. Shah, A. Catellani, H. Reddy, N. Kinsey, V. M. Shalaev, A. Boltasseva, and A. Calzolari, "Controlling the Plasmonic Properties of Ultrathin TiN Films at the Atomic Level," ACS Photonics 5 (7), 2018, and D. Shah, H. Reddy, N. Kinsey, V. M. Shalaev, and A. Boltasseva, "Optical Properties of Plasmonic Ultrathin TiN Films," Adv. Optical Mater. 5 (13), 2017.

1.1 Optical Properties of TDMs

Several theoretical studies have revealed the strong thickness dependence of plasmonic films due to quantum size effects and their potential use in low loss plasmonics[35]–[43], many of them depending on density functional theory (DFT) calculations. As the number of

layers of the Au and Al films is reduced, the complex permittivity becomes highly anisotropic at longer wavelengths and also exhibit a decrease in losses[35], [37]. In the case of ultrathin plasmonic TiN, we show that reducing the number of layers results in red shifts in the interband transitions causing changes in the optical properties[41]. The results from this study is discussed in further detail in Chapter 3. Additionally, monolayer Ag has been predicted to have a higher frequency dependent conductivity and lower losses than its bulk counterpart[42].

Along with DFT methods, there have also been analytical solutions proposed to describe the evolution of optical properties in ultra-thin metals and their nonlocal response. Quantum mechanical (QM) models have been developed by applying the solution of the Schrödinger equation to atomically thin metal layers[39], [43]. Furthermore, Echarri et al. applied the formulated QM models to graphene-metal heterostructures to demonstrate the quantum size effects on the plasmon dispersion and lifetimes as well as the viability of acoustic plasmons with monolayer gold and graphene systems[43].

A nonlocal Drude response model has also been developed based on the Keldysh – Rytova (KR) potential, which describes the Coulomb interaction between electrons in transdimensional plasmonic planar films[40]. The derived model provides a mathematical description of the nonlocal effects that result in the thickness dependent complex permittivity and Drude plasma frequency of ultra-thin plasmonic films. Using this nonlocal KR model, many remarkable electromagnetic (EM) properties of plasmonic TDMs have also been demonstrated. For example, the strong spatial confinement of the electron transport in TD films has been predicted to result in a resonance magneto-optical response. Moreover, when carefully tuned, the ultra-thin films of finite lateral size can become negatively refractive in the IR frequency range[44]. The nonlocal KR model has also been used to study the coupling between the ENZ modes in TDMs and dipole emitters, revealing large dipolar spontaneous emission enhancement, which can be precisely controlled by the thickness dependent plasma frequency of the TD films[45].

1.2 Strongly Confined Plasmons in TDMs

Highly doped graphene has long been established as a promising 2D material platform that can combine huge confinement and enhancement of the electromagnetic field with lower losses and large electrical modulation of its optical properties[17], [27], [46]. However, due to their lower carrier densities, the strongly confined plasmons in graphene, as well as other 2D materials and topological insulators, lie at lower frequencies. In contrast, atomically thin metals support confined plasmons in the technologically important visible and NIR range because of their larger charge carrier density. As observed with plasmons in graphene, decreasing the thickness d of metallic films down to a range of just few atomic layers, results in a significant reduction of the in-plane surface-plasmon wavelength λ_p . In the Drude model, λ_p is related to d and the wavelength of light λ_0 as

$$\lambda_p = \frac{d\lambda_0^2}{L_1^2}$$

where L_1 is a characteristic length that depends on the combination of metal and substrate materials[47].

Advancements in thin film growth have resulted in many material choices to study these confinement effects on the plasmonic behavior in atomically thin metals and characterize the strongly confined plasmons in the visible and NIR in TD plasmonic structures. Quantum well states have been characterized in epitaxial silver films of 10 monolayers[47]. The layer dependent quantum well binding energies reveal trends characteristic of a 1-D particle in a box. In terms of the plasmonic properties, narrow, strongly confined plasmons at NIR wavelengths were measured in the silver films patterned into nanoribbons. Highly confined low dimensional plasmons have also been characterized in monoatomic gold wires[48] as well as ultrathin, continuous silver thin films[49].

According to a recent theoretical prediction, the transition rates of various forbidden transitions in atomic physics, such as electron spin-flips and multi-plasmon spontaneous emission, can be greatly enhanced by several orders of magnitude by using these strongly confined plasmons in TDMs [30]. Due to the large momentum mismatch between the photon momentum and the orbital angular momentum of the electron, transitions involving changes in the orbital angular momentum of electrons are often considered to be impossible in light-matter interactions. For instance, the transition rates that change the orbital angular momentum of electron by n scale as $\alpha(ka_0)2n$, where α is the fine structure constant, a_0 is the characteristic size of the atom and k is the wavenumber of the photon. Due to the small size of atom a_0 the lifetimes of these transitions are quite large. However, using tightly confined plasmons in ultra-thin plasmonic films, where the plasmon wavenumber is larger than the free space wavenumber by several orders of magnitude, these transition rates can be greatly enhanced by several orders of magnitude, reducing the lifetimes to experimentally observable sub-micron timescales for transitions in the visible and NIR.

1.3 Confinement Enhanced Optical Nonlinearities

In bulk materials, the nonlinear response is dominated by the interband transitions, i.e. when an electron transitions from the d-band (valence band) to the s-band (conduction band). Due to quantum confinement in atomically thin metals, free electrons of the metal are quantized into subbands or discrete energy levels within the conduction band. Therefore, there are also intersubband transitions, which enhance the nonlinear response. The dipole transition elements associated with transitions between intersubbands are on the same order of magnitude of the quantum well thickness: $\mu_{mn} = E(\cos\alpha)\langle e_m|qz|e_n\rangle$, where z is the thickness of the quantum well, which is typically a few nanometers. On the other hand, the dipole transition elements for interband transitions are on the same order of magnitude as the interatomic distance: $\mu_{mn} = E\langle e_m|e_n\rangle\langle u_c|qr|u_v\rangle$, which is typically a few angstroms[50]. The larger dipole transition moment for intersubband transitions leads to a significantly larger $\chi^{(3)}$ value,

$$\chi^{(3)} = \frac{N(\mu_{mn}^4)}{(\omega_{mn} - \omega - \mathrm{i}\gamma_{mn})^3}$$

The addition of new resonant terms in the permittivity due to quantum confinement results in an enhanced nonlinear response. The third order nonlinearities in TiN films have been characterized, and it has previously been shown that such nonlinearities can be increased for thinner films [51]. Specifically, in the Drude tail of the material where there is no nearby resonant term, quantum size effects may result in substantial enhancement and modification of the nonlinearity by allowing energy level quantization, whose spectral location can be tuned with the thickness and/or coupling with other layers to coincide with a wavelength of interest. The ability to engineer the position of such a resonance would be a breakthrough in integrated nonlinear optics, allowing the material to be tailored to the wavelength of interest instead of the wavelength being tailored to the material.

The third order nonlinearities in ultra-thin gold films with thicknesses 3 and 15 nm grown in quartz have been studied using Z-scan measurements[34]. They demonstrate that the 3 nm thick gold quantum well has a $\chi^{(3)}$ value that is four orders of magnitude larger than the ones for bulk, 15 nm, and 20 nm films. Additionally, due to some weak confinement, the 15 nm and 20 nm films are slightly larger than that for bulk. The study provides further evidence to show that the enhanced nonlinearity is a purely quantum effect by measuring the wavelength dependent Kerr response, demonstrating that the resonances in the Kerr response (740 nm and 900 nm), correspond to the energies required for intersubband transitions in the metallic quantum well.

In addition to enhanced third order nonlinearities, TD metallic films may also be utilized to achieve enhanced second harmonic generation [52]. Enhancement of the second harmonic generation using asymmetric semiconductor quantum wells has been widely demonstrated with wavelengths in the mid infrared range. Using metallic TDMs, significant enhancement can be realized in the visible to NIR range. Multilayers of TiN and alumina, which are essentially asymmetric coupled metallic quantum wells, have been introduced to realize extremely high $\chi^{(2)}$ values at NIR wavelengths. The $\chi^{(2)}$ value of the TiN/Al₂O₃ superlattice reported is several orders of magnitude larger than traditional plasmonic structures. By combining the quantum well heterostructure with the large electric field enhancement offered by a nanostructured plasmonic metasurface, at a pulse intensity of 10 GW/cm², a power efficiency of 10⁴ for second harmonic generation (SHG) was achieved. This efficiency is higher by several orders of magnitude compared to previous studies investigating the second order nonlinearity of different structures at similar wavelengths.

1.4 Tuning the Properties of Metals

Transdimensional plasmonic films are predicted to have a greatly increased sensitivity to the local dielectric environment, strain, and external perturbations. Consequently, unlike conventional metals with properties that are challenging to tailor, atomically thin plasmonic materials exhibit optical responses that can be significantly engineered by precise control of their thickness, composition, and the electronic and structural properties of the substrate and superstrate[33], [40], [53]. This unique tailorability, unachievable with bulk or relatively thick metallic layers, establishes metallic TD films as an attractive material platform for the design of tunable and dynamically switchable plasmonic metasurfaces[54].

The tailorable plasmonic behavior of atomically thin gold and silver has been demonstrated by careful control of the thickness. In an experimental study with gold, the behavior of the plasmon dispersion in ultra-thin gold nanoribbons with width W and thickness of 3 nm deviated significantly from those with a 15 nm thickness. In a separate study, considerable redshifts in narrow, strongly confined plasmons at NIR wavelengths were measured in epitaxial silver nanoribbons with thicknesses ranging from 8 - 12 monolayers. In this thesis, we have focused on demonstrating the thickness dependent optical properties in TiN. Moreover, recently, a nonlocal Drude response model was derived to theoretically demonstrate that the plasma frequency of ultrathin metals acquires a thickness dependence, which will be discussed in Chapter 4 in terms of the experimental trends we observe in TiN.

Strain engineering is also a critical method to understand and control the optical behavior, as it enables modification of the energy band curvature and related quantities[53]. Below a certain critical thickness, an epitaxial thin film will retain strain induced by the substrate. The effects of strain are only felt for of a very thin layer of an epitaxial material near the interface of the super and substrate, as the stress relaxes for atoms farther from the interface. Thus, to properly quantify the effects of strain, it is important to successfully obtain single – crystalline, ultrathin layers, where the strain will uniformly affect the deposited material, drastically simplifying modeling and experimental extraction. In conjunction with the thickness dependence of TDMs, strain engineering epitaxial TD films offers a novel avenue to tailor the optical response of plasmonic materials.

Moreover, similar to 2D materials [55], [56], high sensitivity to external optical and electrical stimuli is expected in plasmonic TDMs, which could facilitate the realization of ultrathin plasmonic devices where the properties can be dynamically modulated [33], [57]. As recently predicted[33], the free carrier densities, and thereby the optical response itself, can be significantly tuned with conventional gating techniques, provided that the metal film is only a few monolayers in thickness. For example, the dynamic modulation of plasmonic resonances in atomically thin metals has been experimentally demonstrated in 3 nm thick gold nanoribbons [58]. A 200 nm wavelength shift in the plasmonic resonance was observed by applying voltage biases from -2 V to 2 V, compared to only a 50 nm wavelength shift in a 6 nm film. The manipulation of the local density of optical states has also been experimentally demonstrated by utilizing the field effect-induced optical permittivity modulation of an ultrathin TiN film in a $TiN/SiO_2/Ag$ heterostructure [54]. As a result, the spontaneous emission rate, far-field emission intensity, and quantum yield of quantum dots embedded in the heterostructure can be modulated. Additionally, active tuning via electrical gating of surface plasmon resonances in transdimensional indium doped tin oxide (ITO), a low carrier density Drude metal, has also been demonstrated in the NIR range [59]. These results indicate strong potential for metallic TDMs as a material platform for dynamic plasmonic and nanophotonic components for switchable metasurfaces.

1.5 Material Platforms for Transdimensional Plasmonics

To explore the sought after transdimensional thickness regime, there have been several efforts to grow ultra-thin metallic films with various plasmonic materials, such as gold[60] and titanium nitride (TiN)[61], [62]. Until recently, experimental investigations of atomically thin plasmonic materials had been largely limited due to the challenges in growing ultra-thin metal films that are continuous and have low roughness. For example, at small thicknesses, thin films of noble metals, which are the typical material platform choice for plasmonic devices, suffer from high roughness and nonuniformity due to island formation[63], [64]. However, due to recent progress in deposition techniques, continuous and smooth thin films of gold, silver and TiN that maintain their metallic properties at thickness of just several

monolayers have become easier to achieve. For instance, wetting layers, such as copper and MoS2, have been utilized to reduce the percolation threshold of noble metals and produce continuous films down to a few nanometers [58], [60], [65], [66]. Other techniques to obtain atomically smooth metal films involve e-beam evaporation of silver onto substrates cooled down to cryogenic temperatures to achieve high quality epitaxial plasmonic films [47], [67].

Epitaxial thin films are particularly beneficial in experimentally investigating the size dependent response of metals as there are no contributions to the optical properties from grain boundaries that would increase the scattering and losses. However, so far there have not been many demonstrations of epitaxial gold films down to thicknesses of just several monolayers. Additionally, while atomically thin epitaxial silver films with thicknesses of just 3 nm have been achieved[47], silver films are extremely chemically unstable and easily degrade when exposed to ambient environments, making it difficult to experimentally study the size dependent plasmonic behavior and realize practical devices with TD silver.

In contrast, titanium nitride (TiN) has been widely studied as an alternative plasmonic material due its chemical and mechanical stability, tailorability, and refractory properties. TiN has already been shown to have potential applications in photovoltaics[68], [69], waveguiding[7], [70], modulators[71], and nonlinear optical devices[51], [72]. Moreover, TiN can be grown epitaxially on lattice matched substrates such as c-sapphire and MgO, facilitating the development of ultra-thin, continuous films down to thicknesses of just 1 nm while maintaining a low roughness and high optical quality. Due to its high-quality growth, chemical and mechanical stability and tailorable optical properties, TiN is the ideal material choice to study the plasmonic properties as TDMs evolve from a few atomic layers to a bulk material. The electronic, optical and mechanical properties of TiN single or polycrystalline bulk system and thick films (>100 nm) have been largely studied[73]–[77] using several experimental and theoretical techniques, including spectroscopic ellipsometry[78], [79], X-ray, UV photoemission and Auger spectroscopy[80], [81], electron energy loss[82], [83], Hall electron transport[84], scanning electron and tunneling microscopies[85], [86], tight-binding and ab initio simulations[73], [87]–[89].

Although few reports on growth[90], [91] and electron transport[92], [93] of ultra-thin TiN films below thicknesses of 10 nm have appeared in the previous years, the optical properties

of TiN films in this thickness range are still largely underexplored. In this thesis, I will present our experimental and theoretical investigations on studying the thickness dependent optical properties of TiN in the TD regime. We experimentally demonstrate that TiN films with thicknesses down to 1 nm remain plasmonic in the visible and NIR wavelength range. Moreover, we use density functional theory (DFT) and a nonlocal Drude response model to theoretically explain the experimental thickness dependent trends observed, which are induced by strong electron confinement in the ultra-thin films. Finally, we explore the possibility of electrically tuning the properties as well as the ultrafast carrier dynamics in TD TiN.

1.6 Outline

This thesis contains the major results from our investigations on the evolution of the optical and plasmonic properties in transdimensional TiN.

Chapter 2 covers our initial studies on the plasmonic properties of unpassivated ultrathin TiN with thicknesses down to 2 nm. We find a strong dependence of the optical response on the thickness, which is attributed to surface oxidation, increase in surface scattering, and quantum confinement effects. Despite the decreasing metallicity in thinner films, the thinnest films (2 nm) remain highly metallic with a high carrier concentration on the order of 10^{22} cm⁻³.

Chapter 3 discusses our results from a joint experimental and theoretical investigation of the electrical and optical properties of ultra-thin TiN to understand the underlying physics that result in their thickness dependent properties. Using density functional theory (DFT) calculations, we reveal that the electron confinement in thinner films causing a redshift in the plasmonic response. We demonstrate that in this TD thickness regime, the thickness, surface oxidation, and substrate properties are crucial factors in controlling the optical behavior.

Chapter 4 introduces a nonlocal Drude response model that describes the confinement induced optical properties in plasmonic TD films. We characterize the optical properties of passivated TD TiN films with thicknesses ranging from 1-10 nm to verify the developed non-

local theory. Our experimental results demonstrate quantum confinement induced thickness dependent optical properties in plasmonic TD TiN.

Chapter 5 discusses the feasibility of electrically tuning the properties of atomically thin TiN. We demonstrate active control of the carrier concentration of a 1 nm TiN thin film. Based on the theoretically achievable carrier concentration modulation, a phase modulator is designed.

2. OPTICAL PROPERTIES OF ULTRA-THIN TITANIUM NITRIDE

Most of the content presented in this chapter is taken verbatim from the paper D. Shah, H. Reddy, N. Kinsey, V. M. Shalaev, and A. Boltasseva, "Optical Properties of Plasmonic Ultrathin TiN Films," Adv. Optical Mater. 5 (13), 2017.

2.1 Introduction

With the continuing developments in nanofabrication techniques, structural dimensions and film thicknesses have been decreasing, which has led researchers to probe and understand light-matter interactions at the nanoscale. As the thickness of films approach just a few monolayers, the strong confinement could lead to quantum effects[94], [95] that could greatly enhance light-matter interactions[39], [96]. In addition, these films are expected to be more sensitive to external electrical[33] and optical perturbations, making them an attractive material platform for realizing dynamic nanophotonic and plasmonic applications. However, a key to realizing these applications is the ability to produce metallic films that are only a few monolayers in thickness, which also exhibit good optical and plasmonic properties.

For conventional plasmonic metals, such as silver and gold, continuous, smooth films with thicknesses lower than 10 nm are challenging to produce because of island formation and large defect concentrations which result from their high surface energy[60], [63], [97]. On the other hand, TiN, can be grown epitaxially on substrates such as c-sapphire and MgO, enabling the formation of continuous, plasmonic, ultra-thin films down to 2 nm[98], [99]. TiN has already been shown to have potential applications inphotovoltaics[68], [69], waveguiding[7], [70], modulators[71], and nonlinear optical devices[51], [72]. Although some of these devices use films of 10 nm, there is an increasing interest in the formation of ultrathin metallic films (<10 nm) which may facilitate additional applications resulting from the strong confinement[30], nonlocal effects[40], and highly sensitive optical response[33] that arise in atomically thin films. To support this, there have been efforts to grow ultrathin TiN films using atomic layer deposition (ALD). However, these films become dielectric below 5 nm and may not be useful for nanophotonic applications[62]. This is because TiN thin films grown using ALD with a TiCl₄ precursor are more prone to Cl impurities, which could result in a degradation of the metallic properties in thinner films [100]-[102].

In the present study, we have grown continuous, epitaxial, ultrathin TiN films that remain plasmonic in the optical range via DC reactive magnetron sputtering. Using sputtering to grow the film decreases the chance of contamination since only Ti, N, and Ar are introduced into the system, which helps maintain the metallic properties in ultrathin films. The complex dielectric permittivity of these films was characterized using spectroscopic ellipsometry, while charge carrier densities extracted from Hall measurements give further insight into the thickness dependent optical behavior of the films.

2.2 Methods

2.2.1 Material Growth

Ultra-thin TiN films with thicknesses of 2, 4, 6, 8, and 10 nm were grown on MgO using DC reactive magnetron sputtering. Due to the lattice matching between TiN and substrates such as MgO and sapphire, high quality, epitaxial films are possible and have been demonstrated using a similar growth technique in past studies[70], [99]. The TiN films were deposited by sputtering a titanium target in a 60% nitrogen and 40% argon environment at 5 mT with 200 W of DC power with the substrate heated to 800° C. A growth rate of 2.2 \pm 0.1 nm/min was determined by depositing an approximately 200 nm thick sample over a fixed deposition time of 100 minutes and then measuring its thickness via SEM, which had a 5% error bar. Subsequently, the sputtering time for the desired thickness was calculated and films of desired thicknesses were grown. This growth procedure resulted in high quality, continuous films with low roughness of <0.21 nm, as shown in the atomic force microscopy (AFM) images of the 2 nm, 6 nm, and 8 nm films in Figure 2.1.



Figure 2.1. AFM images of (a) 2 nm, (b) 6 nm, and (c) 8 nm TiN films showing a very low roughness of <0.21 nm, indicating atomically smooth films

2.2.2 Optical Characterization

Ellipsometry measures the change in polarization of the reflected light from the sample's surface. This change is measured in terms of the amplitude ratio, Ψ , and the phase difference Δ ,

$$\rho = \frac{R_p}{R_s} = tan(\Psi) e^{i\Delta}$$
(2.1)

where R_p and R_s are the Fresnel reflection coefficients for p-polarized and s-polarized light respectively. The linear optical properties were measured using variable angle spectroscopic ellipsometry at angles of 50° and 70° for wavelengths from 400 nm to 2000 nm. The measured Ψ and Δ values at 50° and 70° are plotted in Figure 2.2.

The optical constants of TiN were determined by fitting the measured ellipsometry data using a Drude – Lorentz model,

$$\varepsilon(\omega) = \varepsilon_b - \frac{\omega_{pD}^2}{\omega^2 + i\Gamma_D\omega} + \sum_L \frac{f_L \omega_L^2}{\omega_L^2 - \omega^2 - i\Gamma_L\omega}$$
(2.2)

where ε_b is the high frequency background permittivity, ω_{pD} is the Drude plasma frequency, Γ_D is the Drude damping due to inelastic electron scattering, f_L , ω_L , and Γ_L are the strength, resonant frequency and broadening of the Lorentz oscillator, respectively. The



Figure 2.2. Ellipsometry measurements, Ψ and Δ , of the ultrathin films. (a) and (c) are Ψ and Δ for 50°, while (b) and (d) are Ψ and Δ for 70°

Drude term captures the optical response of the free carriers, while the Lorentz term accounts for interband transitions.

The real (ε') and imaginary (ε'') parts of the Drude oscillator in the complex dielectric permittivity in Eqn. (2.1) can be written as,

$$\varepsilon'(\omega) = \varepsilon_b - \frac{\omega_{pD}^2}{\omega^2 + \Gamma_D^2} \tag{2.3}$$

$$\varepsilon''(\omega) = \frac{\omega_{pD}^2 \Gamma_D}{\omega^2 (\omega^2 + \Gamma_D^2)}$$
(2.4)

The Drude plasma frequency ω_{pD} is given by

$$\omega_{pD} = \sqrt{\frac{Ne^2}{m^* m_e \varepsilon_0}} \tag{2.5}$$

where N is the charge carrier density, ε_0 is the permittivity of free space, m_e is the mass of an electron, and m^* is the electron effective mass. The real part of the complex permittivity represents the strength of the polarization induced by an external electric field. It also signifies how metallic the plasmonic film since it is proportional to the charge carrier density through the Drude plasma frequency. The imaginary part of the complex permittivity describes the optical losses encountered in the material.

In atomically thin films, the strong confinement in thickness leads to the emergence of quantum phenomena, which may not be accurately captured by a simple Drude-Lorentz model. However, the thinnest film considered in the current study (2 nm) is nearly five monolayers in thickness. A recent computational work on atomically thin gold films[33] revealed that the optical properties described by a simple classical free electron model match well with quantum mechanical calculations. It is therefore reasonable to describe the dielectric function of all the presented films with a Drude-Lorentz model, given that they are several monolayers in thickness. For even thinner films, quantum size effects may lead to deviations in the Drude-Lorentz model, but these studies are beyond the scope of current work.

2.2.3 Electrical Characterization

The carrier densities in the TiN films were measured using standard Hall measurements. Using photolithography and a dry etching process, Greek crosses were fabricated from the TiN films and gold contact pads were applied. Positive resist AZ-1518 was spin coated onto ultrathin TiN films at 4000 rpm. The coated samples were then soft baked at 100 °C for 55 s. Subsequently, the Greek crosses were patterned using the Karl Suss MJB3 Mask Aligner and developed in MF 26A for 40 sec. The developed samples were then dry etched (Panasonic E620) using an optimized chlorine etch recipe, producing the Greek crosses. Finally, 180 nm thick gold contact pads were deposited using a lift-off technique. A 6 nm Ti layer was used

as an adhesive layer during deposition. Hall measurements of planar materials are more susceptible to error due to a finite contact size. Subsequently, the influence of the finite contacts on the measurements can be mitigated by using a Greek cross geometry (Figure 2.3)[103]. Figure 2.3 shows an optical image of a fabricated Greek cross on an 8 nm TiN film. The Hall coefficient was determined by measuring the voltage across two arms of the cross while a current (10 mA) passing through the other two arms generates an out-of-plane magnetic field. Because the relationship between the Hall resistance R_H and the applied magnetic field B is linear,

$$B = N de R_H \tag{2.6}$$

where N is the carrier density and d is the thickness of the film, the carrier density N can be extracted.



Figure 2.3. Optical image of the an 8 nm Hall cross with the gold contact pads used for extracting the charge carrier densities. The Hall cross was fabricated using photolithography and a dry etching process.

2.3 Results

2.3.1 Optical Properties of Ultrathin TiN

The Drude – Lorentz model parameters that describe the complex dielectric permittivity per Eqn. 2.2 are given in Table 2.1, and the real and imaginary parts are plotted in Figure 2.4a and 2.4b, respectively. As the thickness decreases, the films are observed to become less metallic as signified by the decrease in the magnitude of the real part of the dielectric permittivity. This is consistent with the results from the Hall measurements (Figure 2.5) which show that the free electron concentration decreases along with film thickness. The decreasing carrier concentration can be attributed to an increase in the films' defect concentration caused by the presence of a 2 nm layer of $\text{TiN}_x O_y$ that is typically formed on the surface of TiN films[104], [105]. In thinner films, due to a larger surface to volume ratio, the relative oxygen concentration increases, resulting in a reduced free carrier density. However, we note that the thinnest films (2 nm) still have a carrier density on the order of 10^{22} cm^{-3} , demonstrating that they retain their metallic character.

Thickness (nm)	2	4	6	8	10
$\varepsilon_{\mathrm inf}$	0.92	2.66	2.95	3.11	3.28
$\omega_{pD}^2~({ m eV}^2)$	34.98	44.97	48.2	50.2	51.66
Γ_D (eV)	0.282	0.247	0.236	0.232	0.229
$f_1\omega_1^2 \ ({ m eV}^2)$	0.28	0.58	0.7	0.68	0.72
$\Gamma_1 (eV)$	0.35	0.45	0.5	0.49	0.53
$\omega_1 \ (eV)$	2.15	2.16	2.18	2.19	2.16
$f_2\omega_2^2~({ m eV}^2)$	113.9	105.2	100.1	99.6	96.7
$\Gamma_2 (eV)$	2.38	2.48	2.47	2.54	2.53
$\omega_1 \text{ (eV)}$	4.44	4.65	4.7	4.78	4.79

Table 2.1. Ellipsometry fitting parameters for ultra-thin TiN using a Drude- Lorentz model



Figure 2.4. Thickness dependent (a) real and (b) imaginary parts of the dielectric permittivity as retrieved from ellipsometry measurements. The films become less metallic with decreasing thickness.

Additionally, at visible wavelengths, we see an increase in the imaginary part of the permittivity of thinner films, while there is a decrease in thinner films in the near infrared. According to Eqn. 2.4, ε'' is quadratically proportional to the Drude plasma frequency and linearly proportional to the Drude damping. In order to understand the experimental trends for ε'' , I will first discuss the thickness dependent trends of the Drude damping and plasma frequency that were retrieved from the ellipsometry fittings.



Figure 2.5. Carrier densities obtained from Hall measurements and damping factor from ellipsometry measurements. As the thickness decreases, the metallic character decreases and the scattering losses increase. The error bar for the carrier densities is obtained from the error in the sputtering rate.

2.3.2 Scattering Losses in Ultrathin TiN

Additionally, in Fig. 2.5 we see an increase in the Drude damping factor for thinner films, as obtained from ellipsometry. In general, there are five scattering processes which produce the losses in thin films, which include bulk scattering due to thermal lattice vibrations, electron-electron scattering, surface scattering, defect scattering, and grain boundary scattering. Since the films are epitaxial, scattering at grain boundaries can be neglected. On the other hand, as the film thickness decreases, both defect and surface scattering become important contributors to losses[106]. When the film thickness is reduced below the mean free path of the bulk material, the surface scattering increases as[107]

$$\frac{1}{\tau_{ss}} = \frac{5Ne\rho_b\lambda}{8m^*m_ed}(1-p) \tag{2.7}$$

where N is the carrier concentration, e is the electron charge, m^* is the effective mass, m_e is the electron mass, ρ_b is the bulk resistivity, λ is the electron mean free path, d is the film thickness, and p is a scattering parameter. The scattering parameter, p, may vary from 1 for complete specular scattering to 0 for complete diffuse scattering. For the case of bulk titanium nitride, the mean free path is approximately 45 nm[92], which is considerably greater than the film thicknesses studied such that surface scattering is expected to be significant. Additionally, the defect concentration is expected to increase for thinner films. Since the oxygen defects are primarily located within 2 nm from the surface, a reduction in the thickness results in a higher overall density of defects and subsequently an increased scattering probability.

2.3.3 Thickness Dependent Drude Plasma Frequency

The plasma frequency, ω_{pD} , of each film as extracted from the fitted ellipsometry models (with the error ranging from 2% to 8%) is also observed to decrease with film thicknesses as plotted in Figure 2.6. The plasma frequency is dependent on the carrier concentration and the effective mass of the electrons by the definition

$$\omega_{pD} = \sqrt{\frac{Ne^2}{m^* m_e \varepsilon_0}} \tag{2.8}$$

where N is the charge carrier density, ε_0 is the permittivity of free space, m_e is the mass of an electron, and m^* is the electron effective mass. Since the plasma frequency is proportional to the square root of the carrier concentration, ω_{pD} gives a gauge for the metallic character of the films. Consequently, changes in the plasma frequency can be attributed to either the carrier density, which is observed to decrease with thickness in our films, or the electron effective mass.



Figure 2.6. Plasma frequency extracted from the ellipsometry fittings (blue) compared to the plasma frequency calculated assuming an average effective mass (red). Below 4 nm, the calculated plasma frequency is outside of the error bar for the ellipsometry values and begins to deviate from the experimental trend, which may be a result of either modifications in the band structure caused by the oxynitride or quantum confinement effects.

2.3.4 Thickness Dependent ε''

As mentioned earlier, ε'' increases in thinner films in the visible range, while it decreases in thinner films in the near infrared. This can be understood from the experimental trends observed for the Drude damping and Drude plasma frequency, since ε'' is dependent on both of these parameters. We experimentally observe a decrease in the plasma frequency and an increase in the damping factor as the thickness decreases. Both of these trends would have an opposing effect on the thickness dependent ε'' , since the plasma frequency would result in a decrease in the overall response while the damping factor would cause an increase. At
longer wavelengths in the near infrared, we find that the thickness dependent trend of the plasma frequency dominates over the damping factor the resulting optical behavior as ε'' begins to decrease in the thinner films.

2.3.5 Role of Effective Mass in the Thickness Dependent Optical Properties

From Equation 4, the electron effective mass is calculated using the values of ω_{pD} from the ellipsometry and the carrier concentration from the Hall measurements. Subsequently, the effective mass is also found to decrease along with the film thickness, as seen in Figure 4b. However, this trend in the effective mass would produce an increasing trend in the plasma frequency. Consequently, we conclude that the reduction of the carrier concentration dominates the effect on the plasma frequency, resulting in the reduced plasma frequency for thinner films. To assess the significance of the effective mass on the optical properties, the square of the plasma frequency is plotted as a function of the carrier concentration in Figure 4c. In this case, ω_{pD}^2 is linearly proportional to the carrier concentration, implying that in comparison to the carrier concentration, the effective mass is relatively constant. Based on a linear fit of the plot in Figure 4c, an average effective mass of 1.31 is determined. Using this average effective mass, the plasma frequency is then calculated using Equation 4 and the carrier concentrations from the Hall measurements and replotted in Figure 2.6. For film thicknesses of 4 nm to 10 nm, the calculated plasma frequency falls within the error bar of the experimental values. However, below 4 nm, the model begins to breakdown, and we see a deviation from the experimental values for the 2 nm film. In this case, when the thickness is decreased to 2 nm, we expect a predominantly TiN_xO_y material. Consequently, the oxygen defects in the film may produce strain in the film which modifies the conduction band curvature and hence, the effective mass. Additionally, quantum confinement may arise in films with thicknesses less than 10 nm, as they do in semiconductor quantum wells, and may also lead to modification of the effective mass. Such effects may also explain the deviation in the optical properties experienced for thicknesses below 4 nm [108], [109]although further investigations are required to quantify the role quantum effects play in ultrathin TiN films. Furthermore, the small lattice mismatch between MgO (4.21 Å) and TiN (4.24 Å) invariably introduces strain into the TiN films which would result in a modified effective mass. However, with increasing thickness, the internal strain slowly fades away and the band structure approaches that of bulk. Based on our experimental findings, we conclude that the threshold thickness below which the band structure begins to deviate due to these effects to be 4 nm.



Figure 2.7. (a) Calculated effective mass using Hall measurements and ellipsometry results. (b) Plasma frequency squared as a function of carrier density, which is observed to be linearly proportional, indicating that the effective mass may be relatively constant. The slope of (b) is used to calculate the average effective mass.

2.3.6 Thickness Dependent Resistivity

The resistivity measured from the optical and electrical characterizations are compared. The optical resistivity of the films is calculated using the Drude damping factor and plasma frequency obtained from the ellipsometry fittings, as

$$\rho = \frac{\Gamma_D}{\varepsilon_0 \omega_{pD}^2} \tag{2.9}$$

The electrical resistivity is measured using the four-point probe method on the ultra-thin films. The resistivity from both electrical and optical measurements are plotted in Figure 2.8.

For both the electrical and optical measurements, the resistivity increases in thinner films and qualitatively follow a similar trend as each other. In the case of the calculations based on the optical measurements, the increase in the Drude damping and the decrease in the plasma frequency for thinner films both contribute to a higher resistivity. The increase in resistivity for both optical and electrical measurements can be explained by the increase in scattering events caused by more surface and defect scattering as well as a decrease in the carrier concentration in thinner films.

Differences in the measured data may be attributed to how the scattering effects contribute to the resistivity calculations in each technique. In ellipsometry, the scattering originates from the electrons' interaction with light. On the other hand, electrical measurements depend on the scattering during the electron's path from one electrode to another.

2.4 Summary

Overcoming the challenge of growing high quality, continuous ultra-thin films that remain metallic is of great importance for practical applications of plasmonic structures. We have developed deposition techniques to grow epitaxial quality, ultra-thin TiN films (<10 nm) with low roughness using DC magnetron sputtering. Using spectroscopic ellipsometry and Hall measurements, we have investigated their optical and electrical properties. Metallic properties are retained in films with thicknesses as low as 2 nm, which exhibit a high carrier concentration and remain plasmonic in the visible and near infrared wavelength range. We observed that with reduced thickness the films become less metallic and more lossy, which is explained by a decrease in the carrier concentration as well as larger number of scattering events caused by higher surface and defect scattering. Based on the relationship between the plasma frequency, carrier concentration, and effective mass, it is determined that the reduction in carrier concentration plays the main role in defining the film's optical properties,



Figure 2.8. Optical and electrical resistivity retrieved from ellipsometry and four-point probe measurements, respectively. Increases in the resistivity is due to the lower carrier concentration due to oxidation and a higher losses due to more surface scattering in thinner films as a result of Differences in the calculated resistivity is due to the different scattering effects that contribute to each technique.

as variations in the effective mass are found to be minimal for thicknesses above 4 nm. For the 2 nm film and below, theoretical calculations begin to deviate from experimental values.

3. FIRST PRINCIPLE STUDIES OF ULTRATHIN TITANIUM NITRIDE

Most of the content presented in this chapter is taken verbatim from the paper D. Shah, A. Catellani, H. Reddy, N. Kinsey, V. M. Shalaev, A. Boltasseva, and A. Calzolari, "Controlling the Plasmonic Properties of Ultrathin TiN Films at the Atomic Level," ACS Photonics 5 (7), 2018.

3.1 Introduction

While we've experimentally demonstrated the retention of high metallicity and changes in the optical properties in ultra-thin TiN films[41], a comprehensive investigation of the underlying physics affecting the plasmonic properties in ultra-thin metal films is mandatory to devise processes and control the film properties. Hence, to move towards the realization of practical devices utilizing ultrathin plasmonic materials, additional studies are required to understand the optical behavior of ultrathin films and the different factors contributing to their thickness dependent properties. In this chapter, we present a joint theoreticalexperimental study on the plasmonic properties of ultra-thin TiN films. We have conducted a detailed theoretical and experimental study on the plasmonic behavior of ultrathin (down to a few atomic layers) TiN films to determine the role of thickness, surface oxidation and interface strain on TiN optical properties. Structural and compositional characterization of the films provide further insight into the possible avenues to tailor the optical response. We show that the effects of oxidation and strain on the optoelectronic and plasmonic properties of the thin films emerge as fundamental parameters to optimize the nanostructure response.

3.2 Methods

3.2.1 Density Functional Theory (DFT)

First principles density functional (DFT) calculations are carried out with semilocal (PBE[110]) exchange-correlation functional, using plane wave basis set (with a cutoff energy of 200 Ry) and norm-converging pseudopotentials[111], as implemented in Quantum

Espresso[112] codes. A uniform (28x28) k-point grid is used for summations over the 2D Brillouin zone.

Extended TiN films are simulated using periodically repeated supercells. Each unit cell (4.24 x 4.24 x 55.00)³ has a $(\sqrt{2} \text{ x } \sqrt{2})$ 2D lateral periodicity and contains 1 to 10, 15 and 20 layers of TiN(001), as shown in Figure 1a. The lattice parameter of TiN surface (4.24 Å) is obtained from the optimization of the corresponding bulk crystal[90]. When considering the effect of oxidation, O adatoms are symmetrically adsorbed on each surface to avoid a spurious dipolar field across the cell. To avoid interaction between adjacent slab replicas, these are separated by at least 15.0 Å of vacuum. Each structure is fully relaxed until forces on all atoms become lower than 0.03 eV/Å.

The complex dielectric function $\varepsilon = \varepsilon' + i\varepsilon''$ is evaluated using the code epsilon.x, also included in the Quantum ESPRESSO suite. This code implements an independent particle formulation of the frequency-dependent (ω) Drude–Lorentz model for solids[111], [113]:

$$\hat{\varepsilon}(\omega) = \varepsilon' + i\varepsilon'' = 1 - \sum_{\mathbf{k},n} f_{\mathbf{k}}^{n,n} \frac{\omega_p^2}{\omega^2 + i\eta\omega} + \sum_{\mathbf{k},n\neq n'} f_{\mathbf{k}}^{n,n'} \frac{\omega_p^2}{\omega_{\mathbf{k},n,n'}^2 - \omega^2 - i\gamma\omega}$$
(3.1)

where $\omega_p = \sqrt{\frac{e^2 n_e}{m^* \varepsilon_0}}$ is the bulk plasma frequency, e is the electron charge, n_e the electron density, ε_0 the dielectric permittivity of vacuum and m^* the electron effective mass. $\hbar\omega_{\mathbf{k},n,n'} = E_{\mathbf{k},n} - E_{\mathbf{k},n'}$ are the vertical band-to-band transition energies between occupied and empty Bloch states, labelled by the quantum numbers (\mathbf{k}, n) and (\mathbf{k}, n') , respectively. $f_{\mathbf{k}}^{n,n}$ and $f_{\mathbf{k}}^{n,n'}$ are the oscillator strengths for the Drude and Lorentz parts and are related to the dipole matrix elements between Bloch states. η and γ are the Drude-like and Lorentzlike relaxation terms, which account for the finite lifetime of the electronic excitations and implicitly include the effects of the dissipative scattering. The capability of the present approach in simulating the optical properties of plasmonic materials has been extensively proved in Refs.[89], [113], [114]. As a further accuracy test, we report in the Supporting Information (SI) the comparison between the dielectric function of TiN bulk (B1 phase) calculated with Drude-Lorentz approach and with the one based on Time-Dependent Density Functional Perturbation Theory, adopted in Ref [89]. The free carrier density ne is obtained inverting the Drude-like expression for the plasma frequency, $n_e = (\omega_p^2 \varepsilon_0 m^*)/e^2$. Rigorously, since TiN has a non-spherical Fermi surface with three bands crossing the Fermi energy, m^* is direction and band dependent. On the other hand, it is an experimental custom to assume a single parabolic model and extract an average m^* (usually deriving from optical measurements) to combine it with resistivity measurements in order to estimate the experimental carrier density. For bulk systems the estimated m^* is very close to the free electron mass $m_0(m^* = 1.1m_0)[84]$. For films thicker than 2 nm the average effective mass is about $1.3m_0$ and it is expected to decrease for thinner films[61]. Direct evaluation from our DFT band structure calculation around point gives $m^* = 0.99m_0$ for bulk and $m^* = 0.9m_0$ for 4L film, respectively, in good agreement with experimental estimates. In the following, for the calculation of n_e we assumed $m^* = m_0$ for each system, with an error band (orange shaded area, Figure 1) among the extrema $m^* = 0.9 - 1.3m_0$.

3.2.2 Material Growth

Ultrathin TiN films with thicknesses of 2, 4, 30, and 100 nm were grown on MgO using DC reactive magnetron sputtering. The TiN deposition was completed by sputter deposition of a titanium target in a 60% nitrogen and 40% argon environment at 5 mT with 200 W of DC power with the substrate heated to a temperature of 800° C, resulting in high quality epitaxial films with low roughness[61].

3.2.3 X-ray Photoelectron Spectroscopy

The Kratos X-ray photoelectron spectrometer was used for the surface analysis of a 30 nm TiN film. A monochromatic Al K- α (1486.6 eV) X-ray source was used as the incident radiation. The wide scan for the overall spectra is performed with a scanning step of 1000 meV, while the narrow scans for the elemental peaks are performed with a scanning steps of 50 meV. The narrow scans were peak fitted using the CasaXPS software. The spectral energy scale was calibrated with the binding energy of C 1s set to 284.5 eV.

3.2.4 Optical Characterization

The linear optical properties of the 2, 4, and 100 nm films were measured using variable angle spectroscopic ellipsometry at angles of 50° and 70° for wavelengths from 400 nm to 2000 nm. A Drude-Lorentz model consisting of one Drude oscillator and one Lorentz oscillator was used to fit the measurements.

3.3 Results and Discussion

3.3.1 DFT Modelling of the Electrical and Optical Properties of Ultrathin TiN

The electrical and optical properties of ultrathin TiN with thicknesses ranging from one to twenty monolayers (0-4 nm) calculated using DFT are summarized in Figures 3.1 and 3.2. Figure 3.1b) reports the thickness variation of the free electron density $n_{\rm e}$. Figure 3.1c) displays the thickness variation of the volume-plasmon energy $E_p = E(\omega_p)$ and of the screened-plasmon energy $E_0 = E(\omega_0)$, where $\omega_0 = 2\pi c/\lambda 0$, and λ_0 corresponds to the wavelength where the real part (ε') of the dielectric function changes sign ($\varepsilon'(\lambda_0) = 0$). Figure 3.2 shows the plots of the (ε') function as resulting from the solution of Drude-Lorentz model of Eqn. (3.1).

All systems exhibit a metallic character and, except for the single layer (1L), have a high carrier density of the order of 10^{22} cm⁻³, in agreement with the experimental results[61] that were presented in Chapter 2. Nonetheless, with decreasing thickness, the films become less metallic. The electron density ne decays almost linearly with the number of layers NL (Figure 3.1b) and the real part of the dielectric function (Figure 3.2) decreases in magnitude as a function of NL (the imaginary part also decreases, not shown). The latter is an indication of the progressive reduction of the electronic screening in ultrathin layers. Additionally, the crossover wavelength 0 red shifts as the thickness decreases. This trend is in qualitative agreement with the experimental results summarized in Chapter 2, where the optical properties of epitaxial ultrathin TiN films with thicknesses of 2, 4, 6, 8, and 10 nm grown on MgO were measured using variable angle spectroscopic ellipsometry.



Figure 3.1. (a) Side view of 1-10, 15 and 20 layer thick TiN(100) slabs used in the simulations. (b) free electron charge density $n_{\rm e}$, and (c) volume-plasmon E_p and screened-plasmon E_0 energies for 1-20 layer films. Although the 20 monolayer thick TiN is highly metallic, the bulk limit for the plasmon energies is not yet reached. The orange shaded area estimates the error bar associated with the choice of m^*

It is important to note that, although the theoretical results cover a smaller range of thicknesses (0-4 nm) and directly compare only with the two thinnest experimental samples (2-4 nm), they are still well representative of the trends showed by the experimental results. This is confirmed also by the $1/N_L$ fit of E_0 displayed in Figure 3.3. In order to extrapolate the limit for films of increasing thickness, we plot (Figure 3.3) the screened plasmon energy E0 as a function of $1/N_L$, where N_L is the number of layers in the simulated system. Two distinct regimes are clearly evident: one dominated by quantum confinement effect for very thin films (1-3 layers, open red circle), one (filled circles) for thicker films. This indicates that trends from theoretical results are meaningful for comparison with experimental data even for thicker films. Figure 3.3 shows that, except for 1L-3L systems, all the other results



Figure 3.2. (a) Side view of 1-10, 15 and 20 layer thick TiN(100) slabs used in the simulations. (b) free electron charge density $n_{\rm e}$, and (c) volume-plasmon E_p and screened-plasmon E_0 energies for 1-20 layer films. Although the 20 monolayer thick TiN is highly metallic, the bulk limit for the plasmon energies is not yet reached. The orange shaded area estimates the error bar associated with the choice of m^*

present a linear behavior as the number of layers $(1/N_L)$ increases (decreases), and the trends can thus be used to compare with experimental data for thicker films.

At the crossover wavelength λ_0 , the imaginary part of the dielectric function ε'' also has a minimum. The condition $\hat{\varepsilon}(\lambda_0) \approx 0$ corresponds to the possibility of exciting a plasmon-like resonance in the visible range[89], [115], [116]. The difference between E_p and E_0 for each thickness value (Figure 3.1c) indicates that the plasmon excitations in the optical range (E_0) do not involve the overall charge density of the system, as for the high energy volume plasmon (E_p) , but only a reduced "screened" fraction $(n_e \approx 10^{21} \text{ cm}^{-3})$. This is the result of a complex interplay between interband and intraband transitions that effectively screen the amount of



Figure 3.3. Screened plasmon energy E_0 as a function of the number of TiN(100) layers (N_L) .

free charge that can be collectively excited in the optical range[90]. E_p and E_0 have two evident functional trends as a function of the thickness: E_p varies in the range 3.1-12.7 eV (1L-20L) with an almost ideal square root behavior (i.e. n_e varies linearly with thickness), while E_0 is limited to a much smaller variation range 1.2-2.9 eV for 1-20 layers, respectively, due to the interband-transition screening that is active even in ultrathin films (see below). We further note that the value of volume-plasmon for TiN bulk is $E_p^{bulk} = 25.4 \text{eV}[82]$, [89], larger than the values resulting in the present case: Despite their metallic character, these films have not yet reached the bulk (or thick-film) density limit, remaining well within the ultrathin regime.

The red shift of the crossover wavelength along with the reduction of the real and imaginary parts of the dielectric function when the thickness is reduced perfectly fits with a recent electromagnetic model explicitly developed to describe the confinement effects on the optical properties of ultrathin plasmonic films[40]. Here, taking advantages of first principles results, we can discuss the effect of quantum confinement in terms of the electronic structure of the ultrathin films. The electron density of states (DOS) of each TiN film close to the Fermi energy (E_F) is characterized by two contiguous groups of bands, labeled **1** and **2** in Figure 3.4, which shows the DOS spectra for the cases 1L (red) and 10L (black).



Figure 3.4. Simulated DOS plots for 1L (red) and 10L (black) systems. Dashed lines remark energy position with respect to the Fermi level (E_F) and the energy width of the two main groups of bands (1, 2) responsible for crossover wavelength in the visible range (λ_0) . Due to electron confinement, a shrinking in the energy bandwidth is observed, resulting in the red shift of the crossover wavelength observed in Figure 3.2

Bands forming the multiplet **1** have a predominant N(2p) character with a minor contribution from $Ti(e_g)$ orbitals, while bands of group **2** mostly derive from $Ti(t_{2g})$ orbitals slightly hybridized with N(2p) states. The plasmonic behavior of TiN originates from intraband transitions of group-2 bands that cross the Fermi level, giving rise to a Drude-like tail in the real part of the dielectric function and to the high-energy volume plasmon. The excitation of interband $\mathbf{1} \to \mathbf{2}$ transitions at lower energies gives a positive contribution to the (negative) Drude component of ε' that becomes globally positive at the crossover frequency λ_0 . Thus, in a first approximation, the energy position of band **1** with respect to the Fermi level (ΔE in Figure 3.4) defines the energy of the screened plasmon (E_0) . The effect of confinement actually changes the energy width and the energy position of bands 1 and 2 with respect to E_F . Despite the two sets of bands are clearly present in both systems, their characteristics are quite different: with decreasing thickness the surface-to-bulk ratio increases along with the number of under-coordinated surface atoms. The presence of frustrated bonds imparts a shrinking of the bandwidth of bands 1 and 2 (i.e. higher spatial electron localization), and an upshift of occupied bands (i.e. lower binding energies), which results in a decrease of ΔE (i.e. E_0 red-shifts). Thus, despite the same formal electron density, the reduction of thickness increases the surface contribution and the electron localization, giving an effective reduction of the free electron charges.

This qualitatively explains the modifications of the optical properties observed in the experiments varying the film thickness[61]. In order to have a more quantitative comparison, we superimpose the experimental ε' spectrum (inset of Figure 3.2) for the 2 nm and 4 nm films. Albeit in qualitative agreement, the theoretical values of 0 are systematically blue shifted with respect to the experimental ones, i.e. the crossover wavelength of 2 nm and 4 nm TiN films are reached at a larger thickness in the experimental studies compared to the theoretical results. In particular, the theoretical value of for the 10L system (i.e. 2nm), $\lambda_0=489$ nm (blue straight line), is blue-shifted by 96 nm with respect to the experimental value, $\lambda_0=585$ nm (blue dashed line). For the 20L system (i.e. 4nm), $\lambda_0=430$ nm (red straight line), is blue-shifted by 105 nm with respect to the experimental value, $\lambda_0=535$ nm (red dashed line). Two different factors that may originate the λ_0 difference between experiment and theory are considered: (i) surface oxidation and (ii) interface strain.

3.3.2 Effect of Surface Oxidation

Typically, a layer of $\text{TiN}_x O_y$ of approximately 1-2 nm is formed on the film surface[104], which would reduce the effective thickness of the films. X-ray photoelectron spectroscopy was conducted on a 30 nm TiN film grown on MgO to study the oxidation mechanisms on the film surface (Figure 3.5). The N_{1s} , O_{1s} , and Ti_{2p} peaks are decomposed as shown in Figures 3.5b, 3.5c, and 3.5d, respectively. Three peaks can be used to fit the N_{1s} spectrum, where the dominant peak at 396.8 eV is attributed to Ti-N bonds. The peak at 397.7 eV is due to the formation of an oxynitride compound, resulting in Ti-O-N bonds[105], [117]. An additional weak peak in the fitting is observed at a lower energy 395.8, which may be the result of other impurities or nitrogen chemically bonded on the surface[105], [118]. While evidence of TiN_xO_y is also present in the O_{1s} spectrum (531.2 eV), the main peak at 529.9 eV is indicative of the existence of Ti-O bonds in the films[119]. These results are also corroborated by the decomposition of the Ti_{2p} peak using three pairs of doublets (2p 1/2 and 2p 3/2) for Ti-N, Ti-N-O, and Ti-O[119], [120] implying that the films are systems consisting of TiN, TiN_xO_y, and TiO₂. The binding energies and shapes of the peaks in Figure 3.5 are consistent with those reported in literature[105], [117], [119], [120].

To determine how the different oxidation phases possibly modify the optical properties, nine different configurations of an oxide capping layer on a 2 nm thick film are considered. These include oxygen atoms in N substitutional sites (O_N) , single oxygen (O_s) and O_2 molecules adsorbed on the surface (Figure 3.6a), and their combination. The corresponding ε' plots are collected in panels b) and c). The dielectric function of ideal 2 nm thick layer (i.e. 10L, black thick line) and the corresponding experimental results (black dashed line) are included for comparison.

Since the metallic bands at the Fermi level are mainly due to Ti-derived states, the coordination of Ti atoms is crucial in the resulting optoelectronic properties of the system. If a single or a few O atoms substitute surface nitrogen atoms, this does not substantially change the Ti coordination leaving the electronic structure almost unchanged (Figure 3.6b). O_N is an aliovalent impurity that acts as an n-type dopant, which increases the electron density. Consequently, a blue shift in the crossover wavelength is observed, contrary to experimental results. Conversely, single oxygen atoms on the surface have little effect on the optical properties (Figure 3.6b). On the contrary, O₂ molecules do cause a red shift of the crossover wavelength, tending towards the experimental value (Figure 3.6c). The O₂ molecules change the coordination of the Ti atoms on the surface to resemble the configuration for TiO₂. As the amount of oxygen is increased, surface Ti easily coordinates with oxygen forming Ti-O bonds at the surface, reducing the overall thickness of pure TiN and leading to the slower blue shift of the crossover wavelengths observed experimentally. To prove this statement we



Figure 3.5. (a) XPS spectrum of a partially oxidized 30nm TiN film grown on MgO by DC magnetron sputtering. The (b) N_{1s} , (c) O_{1s} , and (d) Ti_{2p} peaks are fitted to determine the local chemical environment. CPS are the counts per second collected at each energy. The black curve represents the actual XPS data collected, while the blue dashed curve is the generated envelope curve from peak fitting. The films consist of three phases: TiN, TiN_xO_y , and TiO_2 .

consider, as key example, the case where $3O_N$ and $2O_2$ are included in the system (labeled $3O_N+2O_2$, Figure 3.7). After atomic relaxation, the surface layer displays a remarkable structural rearrangement, where the Ti atoms may have three different kinds of coordination: (i) a partially detached oxide layer forms on the surface, where Ti is bonded to O atoms only. This is in agreement with the Ti-O feature revealed in the XPS data (Figure 3.5c). (ii) In the central region of the slab, Ti is 6-fold coordinated with N, and forms the TiN core of the film (see also main peak in Figure 3.5b). (iii) In the intermediate region N and O coexist and Ti atoms are coordinated with both species giving rise to mixed Ti-N-O phases. This is coherent with the appearance of Ti-N-O features in N_1s and O_1s spectra.



Figure 3.6. (a) Configurations for single oxygen atom in a nitrogen substitutional site (ON), single oxygen atom on the surface (O_s) or diatomic oxygen molecule on the film surface (O_2) . Real part of the dielectric function of oxidized TiN film including (b) single atoms ON and Os; and (c) O_2 molecule. Different combinations of these three main configurations are modeled to determine the effect of oxygen in the films. Theoretical ideal model (10L) and the corresponding experimental spectra are reported in panels b and c for direct comparison.

The effects on the electronic structure are evident in Figure 3.7b, where the comparison with the DOS of ideal 10L system is presented. The total DOS of $3O_N+2O_2$ system is

reminiscent of the original bands 1 - 2 and the system maintains a metallic character, but the presence of oxygen (shaded area) includes new localized states across the Fermi level. This has a twofold effect: first, it reduces the effective thickness of TiN; second, it introduces further O-derived states that contribute to interband transitions, further shifting the crossover wavelength toward red. The oxide layer, however, is not sufficiently thick to exhibit the characteristic dielectric features of TiO₂. Although the exact microscopic reconstruction of the surface strictly depends on the growth process, this example is sufficient to give optical properties in quantitative agreement with experimental findings (Figure 3.6b).



Figure 3.7. (a) Relaxed geometry of $3O_N+2O_2$ configuration. (b) Total (blue line) and O-projected (shaded area) density of states of $3O_N+2O_2$ configuration compared to the ideal unoxidized 10L case (black line). Zero energy reference is set to the Fermi level of the two systems.

3.3.3 Effect of Substrate Induced Strain

Since experimental TiN samples are not free standing but grown on MgO substrate, the effect of a substrate induced strain is included in the simulations and compared to experiments. The lattice mismatch would introduce a strain in the film, changing the in-plane lattice parameter of TiN. In order to investigate this effect without including other structural



Figure 3.8. (a) In-plane strain model. (b) Simulated real part of the dielectric function for 10L TiN films, with varying amount of applied lateral strain. The crossover wavelength red shifts with tensile strain. Theoretical ideal unstrained model (10L) and corresponding experimental spectra are reported in panels (b) for direct comparison.

modifications derived by the explicit description of the substrate, we simulated a set of free standing 10-layer-thick TiN films with varying in-plane lattice constants, which span the range from -3% to 6%, with respect to the equilibrium lattice parameter $(a_0^{TiN}=4.24 \text{ Å})$, as shown in Figure 3.8. As the lattice parameter is reduced (compressive strain), the crossover wavelength blue shifts; the opposite happens when the lattice parameter is increased (tensile strain). To reach the experimental value for the crossover wavelength, a tensile strain of 6% is required. Although ultrathin films may sustain large amounts of strain due to interface mismatch, the latter condition (+6%) does not fit the experimental situation as the actual mismatch between TiN and MgO would result in a compressive strain of only 0.7% (a_0^{MgO} =4.21 Å). Thus, even though in principle strain may affect the optical properties of ultrathin films, in the present case oxidation seems to be the most important cause of the differences between the ideal TiN model and measured sample with the same formal thickness.



Figure 3.9. HAADF STEM image of a 2 nm TiN film on an MgO substrate. The contrast at the film/substrate interface is due to misfit dislocations, causing the film to relax. This indicated that we do not have any significant contributions from strain in the experimentally characterized optical properties

To support these results, the substrate induced strain is experimentally determined by high-angle annular dark field imaging scanning transmission electron microscopy (HAADF STEM) of a 2 nm TiN film grown on MgO, as shown in Figure 3.9. The observed contrast at the TiN/MgO interface signifies the formation of misfit dislocations, indicating that the film has relaxed and has no substrate induced strain which may have influenced the optical properties. The strain relaxation mechanism is typically dependent on the growth process of the material. If the first few monolayers of TiN grown on MgO formed epitaxial islands having the same crystal orientation, the coalescence of the islands would create dislocations contributing to strain relaxation. Once the islands merge, a smooth, continuous, epitaxial film continues to grow[121], [122]. This is consistent with the theoretical findings that the oxidation plays the main role in the difference between the simulations for pure TiN and experimental work. However, as the thickness decreases, the films may remain strained and exhibit a stronger effect on the changes in the optical properties.



3.3.4 Effect of the MgO Substrate

Figure 3.10. Real part of the dielectric function extracted from ellipsometry measurements for ultrathin TiN films grown on MgO and c-sapphire substrate

Although MgO substrate does not affect the position of λ_0 , it acts as a dielectric environment that may change the optical response of TiN films at lower wavelengths ($\lambda < 400$ nm), i.e. where ε' is positive and TiN acts as dielectric medium. In the experimental spectra from ellipsometry measurements, we detect a local maximum at 400nm, which is characteristic only of the 2nm and 4nm films. In the thicker films, the local maximum appears at a lower wavelength (320nm). This is consistent with other experimental and theoretical reports for bulk TiN. At smaller thicknesses, the substrate has a more pronounced effect on the permittivity of the films, which is most probably due to a dielectric effect from the substrate. This is expected to be the cause of the enhancement of ε experimental values compared to the theoretical ones. The same maximum is also present in films grown on c-sapphire (Figure 3.10), which has a comparable refractive index to MgO. The presence of the substrate increases the interband transitions in the UV range between substrate and film, which are responsible for the increase of the dielectric function. MgO does not alter the crossover wavelength of the film, which is strictly dependent on the TiN thickness.



Figure 3.11. Simulated TiN/MgO interface: a) atomic structure (side view); b) real part of the dielectric function; c) total (black area) and TiN- (cyan area) and MgO-projected (red line) density of state. Vertical dashed line marks the Fermi Level (E_F) of the interface.

In order to give a qualitative characterization of the screening effect of MgO substrate on TiN dielectric properties, we calculated the optical properties of a thin TiN/MgO interface. Since TiN and MgO are almost lattice matched, we fixed the lateral periodicity to TiN lattice parameter. The simulation cell, $(4.24 \times 4.24 \times 55.00)^{-3}$, includes 5 layers (1 nm) of TiN(100) and 10 layers (2 nm) of MgO(100), as shown in Figure 3.11a. In order to correct the

well-known bandgap underestimation of standard DFT, we adopted a recent pseudohybrid Hubbard implementation of DFT+U (namely ACBN0 approach [S6]) that profitably corrects the energy bandgap, and the dielectric and vibrational properties of semiconductors [S6]. The U values for Mg and O resulting from ACBN0 cycle and used for the calculations are: $U_{3s}(Mg) = 0.52$ eV and $U_{2p}(O) = 8.30$ eV, respectively. The resulting band gap for MgO bulk is $E_{g} = 7.46$ eV ($E_{g}^{exp} = 7.8$ eV) and the dielectric constant $\varepsilon^{\infty} = 1.8$ ($\varepsilon_{epx}^{\infty} = 2.9$).

Optical properties of TiN/MgO interface are summarized in Figure 3.11b, which shows the real part of the dielectric function (black line), where we superimposed the results for free standing 1nm-TiN (5L, turquoise line) and 3nm-TiN (15L, orange line), for comparison.

Albeit 2nm-thickness is certainly too small to fully reproduce the overall dielectric properties of the MgO substrate, Figure 3.12 underlines interesting issues: (1) the inclusion of the substrate increases the dielectric response of the system in the UV region of the spectrum, which is due to an enhancement of the joint density of states (not-shown) deriving from TiN \rightarrow MgO interband transitions (see DOS superposition in panel c). (2) MgO does not affect the position of the crossover wavelength, which recovers the same value of free standing 1nm-TiN. (3) The blue shift of crossover frequency as the thickness is increased is mainly associated to the thickness of TiN films and not to the overall interface (i.e. to the substrate). (4) The present model reproduces a perfect interface between perfectly matched surfaces. In this case Ti atoms at the interface maintain their original coordination and do not exhibit the red-shift behavior due to oxynitride compounds.

3.4 Summary

We have conducted a detailed theoretical and experimental study on the plasmonic behavior of ultrathin (down to a few atomic layers) TiN films to determine the role of thickness, surface oxidation and interface strain on TiN optical properties. As the thickness is reduced, the metallic character is also decreased because of quantum confinement effects, which increase the contribution from under-coordinated surface atoms leading to an increased electron localization. The atomistic analysis of the experimentally grown TiN thin films indicates a non-negligible deviation from the ideal TiN structure on the surface. We show that the fabricated films' composition can be described as a combination of three different phases: titanium nitride, titanium oxynitride, and titanium dioxide. The surface oxidation of the films, particularly affecting the Ti coordination, has a significant influence on the optical properties of the TiN films, which red shifts the crossover wavelength with respect to the ideal case with the same formal thickness. Additionally, although there is no strain observed in the current films studied, thinner films may exhibit a change in the lattice parameter, which would further affect the optical properties. The investigated ultrathin films remain highly metallic and are characterized by plasmonic excitations in the visible wavelength range. More importantly, the fabricated ultrathin films' optical response can be adjusted by controlling their structural and compositional parameters, such as thickness, strain, and oxidation during growth. The observed plasmonic properties, in combination with the confinement effects, make TiN ultrathin films a promising material for the realization of plasmonic metasurfaces with enhanced nonlinearities and electrical tunability.

4. THICKNESS DEPENDENT PLASMONIC PROPERTIES OF PASSIVATED TRANSDIMENSIONAL TITANIUM NITRIDE 4.1 Introduction

To explore the full potential of metallic TDMs for nanophotonic applications, it is critical to gain insights into the role of confinement and dimensionality on light-matter interactions in atomically thin metals. Recently, theoretical studies devoted to elucidating the confinement-related effects in plasmonic TDMs have revealed the thickness dependent optical response and plasmon dispersion of various metals[35]–[37], [40], [41], [123]–[125]. Relevant to our work, a confinement induced nonlocal Drude dielectric response model based on the Keldysh-Rytova (KR) pairwise electron interaction potential was developed to analytically describe the optical properties in ultrathin metal films as they transition from TDMs to 3D materials[40]. Using this nonlocal KR model, remarkable electromagnetic (EM) properties of plasmonic TDMs have been demonstrated, including large dipolar spontaneous emission enhancement[45], resonance magneto-optical response[44], pronounced negative refraction features[44], [126], and low-temperature plasma frequency drop-off[127].

Until recently, experimental progress in studying the thickness-dependent properties of plasmonic TDMs and validating new theories had been limited due to the challenges in growing ultra-smooth, continuous metallic films with thicknesses below 10 nm[60], [64]. The latest developments in material growth techniques have made it possible to achieve continuous atomically thin metals with controlled thickness and low roughness[47], [65]–[67] [64], allowing one to probe the plasmonic properties of TDMs and light-matter coupling as they evolve from a few atomic layers to a bulk material. Among these high quality plasmonic materials, TiN has been established as a promising material platform to investigate the optical properties in the transdimensional (TD) region due to its epitaxial growth[41], [99], [128], [129], refractory properties[72], [130]–[133], and tailorability[105], [134]–[136]. In chapter 2, we reported on the optical and electrical properties of TiN films with thicknesses below 10 nm[61]. However, the surface oxide that naturally occurs on TiN has prevented the separation of the effects from oxidation and thickness on the optical response evolution in TD TiN[41], [104]. An in-situ passivation layer of AlScN, which can be grown epitaxially on

TiN[137], can help to mitigate the issue of oxidation. Here, we present the optical properties of passivated ultrathin TiN films with thicknesses ranging from 1 to 10 nm to explore the thicknesses dependent trends without any contribution from the oxidation process. We demonstrate the influence of strong confinement on the optical properties of TD TiN and provide the first experimental verification of the thickness dependent plasma frequency induced by electron confinement as predicted by the nonlocal Drude model based on the KR potential.

4.2 Methods



4.2.1 Material Growth

Figure 4.1. AFM images of a 10 nm AlScN film on (a) 2 nm thick TiN film and (b) 8 nm thick TiN film with a root mean square (RMS) roughness of 2.8 Å and 2.7 Å, respectively. The low roughness indicates uniformly passivated and atomically smooth films.

TiN films with thicknesses ranging from 2 to 10 nm were grown on a MgO substrate using DC reactive magnetron sputtering from PVD Products Inc. The TiN deposition was completed by sputter deposition of a titanium target in an 60% nitrogen and 40% argon environment at 5 mT with 200 W of DC power with the substrate heated to 800° C. An in-situ passivation layer of 10 nm AlScN was deposited on the ultrathin films to protect the TiN film from oxidation. AlScN was chosen as the passivation layer since it can be grown epitaxially on TiN[99], [137]. AlScN was grown by co-sputtering Al and Sc targets in a 60% nitrogen and 40% argon environment at 5 mT at a temperature of 800° C. The Al and Sc targets were ignited at 200W and 130W of DC power, respectively. Atomic force microscopy (AFM) scans of the passivated TiN films reveal atomically smooth surfaces with low roughness below 3 Å (Figure 4.1), indicating a uniform AlScN passivation layer on the TiN films. Epitaxial, ultrathin TiN films deposited on MgO, a lattice matched substrate, as well as epitaxial AlScN, have previously been demonstrated using these same processes[41], [99], [137].

To determine the growth rate, a thick TiN film was deposited on silicon for 100 minutes. The thickness was measured using scanning electron microscopy (SEM), which had a 5% error bar. Subsequently, the sputtering time for the desired thickness was calculated. As shown in the TEM image (Fig. 4.2), this method results in accurate thicknesses for continuous films down to 1 nm. The high crystalline quality is also evident from the TEM image and diffraction pattern in Fig. 4.2.



Figure 4.2. High-angle annular dark field imaging scanning transmission electron microscopy (HAADF STEM) image of an epitaxial 1 nm (5 monolayers) TiN film on MgO. The inset shows the diffraction pattern revealing the high degree of crystallinity of ultrathin TiN

4.2.2 Optical Characterization

The optical properties of the AlScN/TiN film stacks were measured at room temperature using a variable angle spectroscopic ellipsometer (VASE) from J. A. Woollam Co. Each measurement was done at incident angles of 50° and 70° for a wavelength range of 400 – 2000 nm. The permittivity of the AlScN passivation layer was first determined by conducting ellipsometry measurements on a 10 nm AlScN film deposited on a 30 nm thick TiN sample with known optical properties. The permittivity of the AlScN layer was fitted using a Lorentz model. The resulting optical constants for the 10 nm thick AlScN film (Figure 4.3) were used for the subsequent fittings of the passivated ultrathin TiN films. The optical properties of the MgO substrate used for the fittings is shown in Figure 4.4.



Figure 4.3. Complex permittivity of the AlScN passivation layer used in the ellipsometry fittings for the passivated TiN films



Figure 4.4. Permittivity of the MgO substrate used in the ellipsometry fittings

The optical properties of the TiN films were fitted using a Drude-Lorentz model as discussed in Chapter 2,

$$\varepsilon(\omega) = \varepsilon_b - \frac{\omega_{pD}^2}{\omega^2 + i\Gamma_D\omega} + \sum_L \frac{f_L \omega_L^2}{\omega_L^2 - \omega^2 - i\Gamma_L\omega}$$
(4.1)

Since the dielectric functions of the ultrathin TiN films are heavily dependent on the thickness in this regime, the thickness is kept fixed to the deposited target thickness during the ellipsometry fittings. The permittivity of the AlScN passivation layer is also kept fixed as the previously fitted data from the 10nm AlScN/30nm TiN sample.

4.3 Results and Discussion

4.3.1 Thickness Dependent Complex Dielectric Permittivity

The fitted parameters for the Drude – Lorentz model for each of the film thicknesses are provided in Table 4.1. The thickness dependent complex permittivity of the passivated TiN films measured from ellipsometry is plotted in Figure 4.5(a). With decreasing thickness, the films become less metallic, as signified by the decrease in the absolute value of the negative real part of the permittivity. On the other hand, the imaginary part of the permittivity increases with decreasing thickness in the near infrared range (Fig. 4.5b) because of the increased Drude damping in thinner films.

Thickness (nm)	1	2	4	6	8	10
$\varepsilon_{\mathrm inf}$	13.8	8.1	8.41	6.17	4.52	5.19
$\omega_{pD}^2 ~({\rm eV}^2)$	47.9	50.65	52.8	54.67	55.71	55.9
Γ_D (eV)	0.40	0.30	0.26	0.237	0.229	0.22
$f_1\omega_1^2 \ ({ m eV}^2)$	14.26	11.6	6.3	56.55	83.38	10.3
Γ_1 (eV)	2.39	0.59	0.45	2.42	1.74	2.8
$\omega_1 \ (eV)$	0.61	3.7	4.14	4.67	5.02	2.27
$f_2\omega_2^2~({ m eV}^2)$	0.125	23.93	24.27	12.94	20.54	99
Γ_2 (eV)	0.1	5.14	5.06	4.58	6.44	1.55
$\omega_1 \ (eV)$	3.02	2.65	2.78	1.9	2.9	5.7

 Table 4.1. Drude - Lorentz fitting parameters for passivated TD TiN films

As discussed in Chapter 2, the imaginary part of the complex permittivity, ε'' , is dependent on both of the plasma frequency and the Drude damping factor. We experimentally observe a decrease in the plasma frequency and an increase in the damping factor as the thickness decreases as seen from Table 4.1, which will be discussed in more detail in the subsequent sections. Both of these trends would have an opposing effect on the thickness dependent ε'' , since the plasma frequency would result in a decrease in the overall response while the damping factor would cause an increase. At longer wavelengths in the near infrared, we find that the thickness dependent trend of the Drude damping dominates over the plasma frequency the resulting optical behavior as ε'' begins to increase in the thinner films. This is contrary to what we had originally observed for unpassivated films. Due to the oxidation of the films, the carrier concentration and the Drude plasma frequency was decreased much more significantly compared to the changes in the Drude damping factor, resulting in a decrease in ε'' in thinner films. In contrast, for the passivated films, since there is no contribution from oxidation, the decrease in the Drude plasma frequency is solely an effect of strong electron confinement and does not decrease as much. On the other hand, the thickness dependent Drude damping is quantitatively very similar to what is observed for unpassivated films. Hence, we find that the thickness dependent Drude damping dictates the changes in ε'' .



Figure 4.5. Complex dielectric permittivity for different thicknesses measured by ellipsometry. (a) Real part of the permittivity indicates lower metallicity with decreasing thickness. (b) Imaginary part increases with smaller thickness, indicating higher losses in thinner films.



Figure 4.6. Thickness dependent epsilon-near-zero (ENZ) wavelength, which redshifts in thinner films. The experimental trend is consistent with the findings from the DFT studies presented in Chapter 3.

Additionally, as seen in Fig. 4.6, the epsilon-near-zero (ENZ) wavelength where the real part of the dielectric permittivity crosses zero, redshifts in the thinner films from 490 nm for the 10 nm thick TiN film to 650 nm for the 1 nm thick TiN film. The lower metallicity in the thinner films can be attributed to a size dependence caused by the confinement of electrons in the vertical direction. These effects stem from an increase in spatial confinement, which results in a strong Coulomb interaction between confined charges that is described by the KR potential[40], [138]. The role of the KR potential is further expanded on and discussed in terms of the plasma frequency in the next section. First principle density functional calculations (DFT) of unoxidized atomically thin TiN have also revealed a similar redshift of the ENZ in thinner films[41], as discussed in Chapter 3. The redshift in the DFT calculations

is explained by quantum confinement causing modifications to the energy width and position of electronic bands in the density of states.

Similar trends have been demonstrated with earlier experimental and theoretical studies on ultrathin TiN films[41], [61]. In past experiments, the decreasing metallicity in thinner TiN films was explained by a combination of a size effect and a 2 nm thick surface oxide layer on the TiN films. In addition to the spatial confinement modifying the optical response, the presence of the surface oxide on unpassivated films reduced the effective thickness of pure TiN and contributed to the interband transitions, causing a further redshift in the optical properties and ENZ wavelength. In the current study, however, the influence of oxidation is avoided by passivating the films with AlScN allowing us to directly probe the size dependence.



Figure 4.7. Comparison between the real part of the permittivity for unpassivated (dashed) and passivated (solid) films with thicknesses of 2 (black) and 4 nm (red). Passivated films have a larger $|\varepsilon'|$, indicating a stronger metallic character compared to unpassivated films because of the absence of oxidation effects.

Due to the absence of the surface oxide, passivated ultrathin TiN films are observed to be considerably more metallic than the TiN films without the AlScN overlayer and show an increase in the absolute value of the negative real part of the permittivity (Figure 4.7). The ENZ of the unpassivated 2 nm thick TiN film is redshifted by 50 nm compared to the ENZ of the passivated 2 nm thick TiN. The increase in the metallicity of unoxidized films is consistent with predictions from DFT simulations for atomically thin TiN films, where it was demonstrated that the inclusion of an oxide layer in the calculation redshifts the ENZ wavelength[41].

4.3.2 Drude Plasma Frequency

The Drude plasma frequency, ω_{pD} , as extracted from the ellipsometry fitting, redshifts in thinner films going from 7.47 eV for the 10 nm thick film to 6.92 eV for the 1 nm thick film (Fig. 4.8a). To understand the observed dependence of the plasma frequency on the thickness, we use the confinement induced nonlocal Drude dielectric response model recently developed for plasmonic TDMs with finite thickness [40], [44], [45]. Our nonlocal dielectric response model is based on the KR pairwise electron interaction potential [138], [139] to describe charged elementary excitations and their electrostatic interactions in vertically confined TDMs[140]-[143]. The KR potential accounts for the electrostatic interaction between a pair of charge carriers confined in a layer of thickness d, which is mostly due to the dielectric surroundings rather than by the material of the layer itself. When d is less than the typical in-plane distance between the carriers, the electrostatic interaction potential of the charge carriers inside the layer is no longer of the Coulomb form. Instead, it becomes the 2D KR potential with the vertical coordinate dependence (z-coordinate) replaced by the layer thickness d. The dimensionality of the system is then effectively reduced from 3D to 2D. The only remnant of the z direction is the d parameter in the KR potential that represents the vertical confinement size of the system [140].

For the visible to NIR frequencies (below the interband transition frequencies), the Drude-Lorentz in-plane EM response function of an isotropic plasmonic TDMs in Eqn 4.1 can be written as,

$$\frac{\varepsilon(k,\omega)}{\varepsilon_b} = 1 - \frac{\omega_p(k)^2}{\omega(\omega + \mathrm{i}\Gamma_D)}$$
(4.2)

Here, the Lorentz terms contribute to ε_b , which accounts for the positive ion background screened by inner-shell electrons. The response of the electrons populating the *s*-band is represented by Eqn 4.2. Additionally,

$$\omega_p(k) = \frac{\omega_p^{3D}}{\sqrt{1 + \varepsilon_1 + \varepsilon_2/\varepsilon_b kd}} \tag{4.3}$$

is the in-plane plasma oscillation frequency of the s-band electron gas with the 2D plasmon momentum $k = |\mathbf{k}|$ in the TDM of thickness d sandwiched between the substrate and superstrate with the dielectric constants ε_1 and ε_2 , respectively. The frequency $\omega_p^{3D} = (4\pi e^2 N_{3D}/\varepsilon_b m^*)^{1/2}$ represents the plasma oscillations in bulk (3D) plasmonic material, where N_{3D} and m^* are the volumetric electron density and electron effective mass, respectively. Equation 4.3 is spatially dispersive, or nonlocal, to result in the nonlocal EM response in Eqn. 4.2. It covers both ultrathin $(d \to 0)$ and thick $(d \to \infty)$ film regimes, yielding the \sqrt{k} and k-independent plasma frequency dispersion for the former and the latter, respectively. This contrasts sharply with standard local Drude-type models commonly used to analyze experimental data[33], [96], including the Drude-Lorentz Eqn. 4.1 utilized here for the studied TiN films. The nonlocality of Eqn. 4.3, which is linear in small k, comes from the strong vertical confinement[44], [144]. This first order nonlocality is much stronger than the second order nonlocality originating from the k^2 dependence of the electron pressure term in hydrodynamical Drude models[145], [146].

At finite temperature, plasmons populate the k-states of the reciprocal space according to the Bose-Einstein distribution function. Therefore, to align the k-dependent theoretical model in Eqn 4.2 with the experimental plasma frequency that assumes no k-dependence in Eqn. 4.1, thermal averaging must be performed on Eqn. 4.2. Under continuous lowintensity light illumination, the plasmonic film is in thermal equilibrium. The standard thermal averaging procedure of Eqn. 4.2 reduces to the plasma frequency averaging over kin Eqn. 4.3 that gives:

$$\bar{\omega}_p(d,T) = \frac{\int_0^{k_c} dk k \omega_p(k) n(k)}{\int_0^{k_c} dk k n(k)}, n(k) = \frac{1}{\mathrm{e}^{\hbar \omega_p^{3D}/k_B T} - 1}$$
(4.4)

where the numerator sums up all boson plasma frequency modes with different k and the denominator provides the total number of such modes in the 2D reciprocal k-space. The k-space is bounded above by k_c , the plasmon cutoff parameter. The Bose distribution function n(k) controls the plasmon occupation numbers of the k-states in the reciprocal space at fixed finite T.

Plugging in Eqn. 4.3 in Eqn. 4.4 gives the expression

$$\bar{\omega}_p(d,T) = \omega_p^{3D} \frac{\int_0^1 dt \frac{t^{3/2}}{\sqrt{t+1/a}(e^{\lambda\sqrt{t/(t+1/a)}}-1)}}{\int_0^1 dt \frac{t}{e^{\lambda\sqrt{t/(t+1/a)}}-1)}},$$
(4.5)

where the dimensionless quantities $t = k/k_c$, $\lambda = \hbar \omega_p^{3D}/k_B T$ and $a = \varepsilon_b k_c d/(\varepsilon_1 + \varepsilon_2)$ are introduced for simplicity. This expression was recently used to explain a peculiar plasma frequency T-dependence observed for 100 nm thick TiN films at cryogenic temperatures[129]. In the case of the ultrathin films here, the quantity a is expected to be less than unity. Additionally, the largest contribution to the integrals comes from the region $t \ll 1$. Then, $at \ll 1$ and $\lambda \sqrt{t/(t+1/a)} \ \lambda \sqrt{at} \ll 1$ unless λ is sufficiently large to invert the sign of this inequality (which is the case of cryogenic $T \ll 10$ K reported earlier[127]). In our experiments that are conducted at room-T, this inequality is very likely to be fulfilled as written. Thus, the Taylor series expansion of the exponentials yields

$$\bar{\omega}_p(d) = \frac{\omega_p^{3D}}{2\int_0^1 dt \sqrt{t(t+1/a)}}$$
(4.6)

to the first nonvanishing order in small $\lambda \sqrt{(t/(t+1/a))}$, whereby the *T*-dependence cancels out and the *d*-dependence stays. For ultrathin films at room-*T*, Eqn. 4.4 loses its *T*-dependence and takes the form

$$\bar{\omega}_p(d) = \frac{2C^2 d^2 \omega_p^{3D}}{(1+2Cd)\sqrt{Cd(1+Cd)} - \sinh^{-1}(\sqrt{Cd})}$$
(4.7)

with $C = \varepsilon_b k_c / (\varepsilon_1 + \varepsilon_2)$, to give the \sqrt{d} behavior in the ultrathin regime.

The ellipsometry data can now be understood in terms of Eqn. 4.5. We note that the standard data fitting expression in Eqn. 4.1 used to extract the data points shown in Fig. 4.4a assumes a slightly different definition for the bulk plasma frequency parameter ω_{pD} than ω_p^{3D} in Eqn. 4.3. Therefore, we normalize the ω_{pD} data points in Figure 4.4a by 7.6 eV, which is the bulk plasma frequency of TiN as extrapolated from Fig. 4.4b. The black dashed line in Fig. 4.4b shows the ratio as given by Eqn. 4.5 with C taken to be 7.7 nm⁻¹. We find that the theoretical and experimental *d*-dependences presented in Fig. 4.4b agree quite well. With the fitted C parameter, ε_b can be evaluated. As the surrounding dielectrics are both weakly dispersive, their static permittivities are approximated as those at a wavelength of 650 nm, where the plasmon resonance condition $Re(\varepsilon) = 0$ is realized in the thinnest film. At 650 nm, $\varepsilon_1 = 3.0$ for the MgO substrate and $\varepsilon_2 = 4.6$ for the AlScN superstrate. We use $k_c = 6.4 \text{ nm}^{-1}$ for the plasmon cutoff as reported recently from the comparison of the Drude model with the first-principles calculations of the low-energy plasmon mode in bulk TiN[90]. We obtain $\varepsilon_b = 9.1$ for the background static permittivity of bulk TiN, yielding $\omega_p^{3D}(\text{TiN}) = 7.6/\sqrt{9.1} = 2.5 \text{ eV}$, which is in very good agreement with both first-principles calculations (2.5 eV, Ref. [89]) and with the electron-energy-loss spectroscopy measurements (2.8 eV, Refs.[82], [83]). For comparison, $\varepsilon_b = 9.5$ and 4 were earlier reported for gold[33] and silver[43], respectively, within the standard local Drude model with no confinement induced nonlocality.

The experimentally measured thickness-dependent plasma frequency agrees well with the predictions of the nonlocal Drude model built on the KR pairwise electron interaction potential. The model stems from a confinement induced effect that results in the \sqrt{d} dependence of the plasma frequency. This is a unique microscopic property of plasmonic TDMs that cannot be obtained macroscopically by imposing boundary conditions on the bulk film interfaces. Similar thickness-dependent plasmonic properties have recently been observed experimentally for ultrathin films of silver[47] and gold[58]. They were interpreted in terms of the standard local Drude model that was adapted to incorporate the known 2D plasmon dispersion[47] and yield the \sqrt{d} dependence for the plasma frequency. While this model can be justified for ultrathin plasmonic films only, the proposed nonlocal Drude dielectric
response model can be applied to all thicknesses covering both the TDM regime as well as conventional thin and thick films.



Figure 4.8. (a) The plasma frequency data points extracted from the ellipsometry spectra measured for ultrathin TiN films of different thicknesses. (b) The data point set of (a) normalized. The black dashed line shows the ratio $\bar{\omega}_p(d)/\omega_p^{3D}$ as given by Eqn. 4.5with parameter $C = 7.7 \text{ nm}^{-1}$. The theory agrees very well to the experiments, indicating that our experiments demonstrate confinement induced optical properties in TD TiN

4.3.3 Thickness Effect on Drude Damping

We retrieve the Drude damping factor, which is the relaxation rate of the free electrons, from the ellipsometry measurements and subsequent fitting. For thinner films, an increase



Figure 4.9. Thickness dependent Drude damping factor from ellipsometry measurements (black squares), from calculations (blue dashes), and calculated Fermi velocity, v_f (red dashes). The increase in Drude damping in thinner films is a result of an increase in surface scattering and larger contribution from surface roughness effects.

in the Drude damping factor is observed (Fig. 4.9). The change in the losses of the thinner films is attributed to a decreased effective mean free path caused by an increase in scattering events. The Drude damping is described by the energy dissipation due to free electron scattering via different mechanisms and can be represented by the sum of the contributing processes,

$$\Gamma_D = \Gamma_{e-e} + \Gamma_{e-ph} + \Gamma_{surf} + \Gamma_{def} + \Gamma_g \tag{4.8}$$

where Γ_{e-e} is the electron-electron scattering, Γ_{e-ph} is the electron-phonon scattering, Γ_{surf} is the surface scattering, Γ_g describes the scattering at grain boundaries, and Γ_{def} is the defect scattering. Since the ultrathin TiN films are single crystalline and passivated, the losses due to grain boundaries and defects are negligible. In bulk epitaxial TiN, the losses, Γ_{bulk} , are dominated by the first two processes (Γ_{e-e} and Γ_{e-ph}). In films with thicknesses smaller than the bulk mean free path l_{bulk} , which is approximately 45 nm for TiN[92], [147], surface scattering increases, resulting in a smaller mean free path[148],

$$l_{eff} = d(ln(\frac{l_{bulk}}{d}) + 1) \tag{4.9}$$

Considering the thickness dependent mean free path, the losses due to surface scattering in TDMs can be defined as[74]

$$\Gamma_{surf} = \frac{\hbar v_f}{l_{eff}} \tag{4.10}$$

where the Fermi velocity, v_f , can be evaluated from the free electron gas model,

$$v_f = \frac{\omega_p}{k_c} \tag{4.11}$$

Although the films studied are atomically smooth and have a very low roughness of 2-3 Å, the surface roughness may have a significant contribution to the Drude damping for thicknesses as small as 1 nm. Since the roughness stays the same with decreasing thickness, the effect of the surface roughness is expected to be inversely proportional to the film. Then, the overall Drude damping factor for an ultrathin film can be described as

$$\Gamma_D = \Gamma_{bulk} + \frac{\hbar v_f}{l_{\rm eff}} + \frac{A}{d} \tag{4.12}$$

where A is a fitting constant to account for the surface roughness, Γ_{bulk} is assumed to be 0.2 eV from the ellipsometry fitting of an optically thick film of 100 nm TiN. Eqn. 4.10 is fitted to the experimentally retrieved Drude damping, resulting in A = 0.12 nm eV. Using Eqn. 4.9, 4.11, 4.12 and the thickness dependent plasma frequency, the Fermi velocity and theoretical Drude damping are plotted as a function of thickness in Fig. 4.9. The calculated Drude damping follows the experimental trend reasonably well, indicating that the ellipsometry results can be well explained by the effect of surface scattering in thinner films. Slight disagreement can be ascribed to the constant $k_c = 6.4$ nm⁻¹ (bulk TiN value) we used throughout the entire range of thicknesses. For thinner films, k_c is expected to decrease due to the Fermi surface shrinkage caused by the size reduction. As the thickness approaches the TD regime below the mean free path, surface scattering events increase and begin to play a dominant role in the losses of the ultrathin plasmonic films.

4.3.4 Drude Damping in Unpassivated Films

Unpassivated TiN films grown using DC magnetron sputtering on MgO substrates are atomically smooth with a roughness of around 2.1 Å, as shown in the AFM image in Fig. 4.10. This surface roughness is on par with that of the passivated films with an AlScN overlayer. Hence, the contribution of the surface roughness to the Drude damping in passivated and unpassivated should be similar. The Drude damping of unpassivated films as extracted from ellipsometry is plotted in Fig. 4.11. The thickness dependent Drude damping in unpassivated films does not vary significantly from the Drude damping of passivated films. It follows the same trend of increasing Drude damping with thickness, indicating that surface scattering should be the dominant process that contributes to the losses in ultrathin TiN.



Figure 4.10. AFM image of an unpassivated 4 nm TiN film showing atomically smooth surface with a roughness of 2.1 Å. The surface roughness of unpassivated films is very similar to passivated films, indicating that the contribution of surface scattering to the Drude damping should be similar.



Figure 4.11. Comparison between the thickness dependent Drude damping of passivated (red) and unpassivated (blue) ultrathin TiN films, as extracted from the ellipsometry fittings. Due to similar surface roughness in both types of films, the trends in Drude damping are quantitatively very similar for passivated and unpassivated films.

4.4 Summary

We report on the thickness-dependent optical properties of passivated transdimensional plasmonic TiN with thicknesses down to 1 nm, or 5 atomic layers. The in-situ passivation layer of AlScN helped to decouple the effects of oxidation and electron confinement in the studied TiN films. The ultrathin films become less metallic as the thickness decreases to 1 nm with a redshift in the epsilon-near-zero wavelength because of quantum confinement, which is consistent with previous theoretical studies20,21. We observe an increase in the Drude damping coefficient due to surface scattering and a decrease in the Drude plasma frequency. The passivated 1 nm thick TiN films remain plasmonic in the visible and NIR range with a ENZ wavelength at 650 nm. More importantly, the thickness-dependent plasma frequency retrieved from the ellipsometry measurements matches very well with the confinement induced nonlocal Drude model. In this model, the vertical electron confinement changes the electron-electron Coulomb potential into the much stronger KR potential and results in the \sqrt{d} plasma frequency dependence in the ultrathin film regime. Our experimental demonstration of thickness dependent optical properties in TD plasmonic TiN provides a crucial step in realizing high quality atomically-thin metallic films that exhibit new material functionalities and unique light-matter interactions. The quantum effects that arise in metallic TDMs along with their strong tunability may pave the way to new optical phenomena and novel, atomically-thin, dynamically tunable nanophotonic devices. Specifically, plasmonic TDMs hold immense potential for enabling a new generation of tunable metasurfaces that utilize strong confinement and nonlocal effects.

5. TRANSDIMENSIONAL TITANIUM NITRIDE FOR ACTIVE METASURFACES

Part of the content presented in this chapter is taken verbatim from the paper H. Jiang, H. Reddy, D. Shah, Z. A. Kudyshev, S. Choudhury, D. Wang, Y. Jiang, A. V. Kildishev, "Modulating phase by metasurfaces with gated ultra-thin TiN films", Nanoscale 11 (23), 2019.

5.1 Introduction

Dynamic modulation of the optical response in metals has been a long-standing challenge in plasmonics. In the absence of interband transitions, the optical properties of metals are primarily determined by the free electrons. Typically, the free electron densities in metallic materials (with negative real part of epsilon) are large, of the order of 10^{22} cm⁻³ which is the major hurdle for dynamic modulation of the optical properties. However, as recently predicted [33], the free carrier densities, and thereby the optical response itself, can be significantly tuned with conventional gating techniques, provided that the metal film is only a few monolayers in thickness. To push the realm of electrical control from the far-/mid-IR to the telecommunication and visible wavelengths, plasmonic TDMs, which are more metallic than graphene are required. The key challenge has been to produce continuous films that are only a few monolayers in thickness and retain their plasmonic behavior in the visible and NIR range. Recently, the dynamic modulation of plasmonic resonances in atomically thin metals has been experimentally demonstrated in 3 nm thick gold nanoribbons^[58]. Additionally, as demonstrated in the previous chapters, epitaxial, atomically flat ultra-thin TiN films with thickness down to 1 nm can be routinely grown. Here, we demonstrate that active tuning of the charge carrier concentration in a 1 nm TiN film can be achieved by performing gate dependent Hall measurements. Based on the modulation of the carrier concentration that can achieved, a phase modulator based that can achieve 337° phase modulation at 1.55 μ m on a 1 nm TiN film is designed.

5.2 Electrical Tunability of Atomically Thin TiN

We have performed preliminary calculations to predict the possible electrical tuning of the charge carrier concentration in TD TiN. Using the bulk carrier density, the areal density in ultra-thin films was computed. A doping density of 10^{14} cm⁻² is assumed, which can be routinely achieved with the existing gating technologies. The ratio of this doping density to the areal density gives the relative change as shown in Figure 5.1. Our initial calculations based on the voltage dependent studies indicate that close to 15% tunability can be achieved in a monolayer of TiN with currently available gating techniques It should be noted that the bulk carrier density was used in these calculations. However, in such thin films the carrier density is expected to be lower than bulk, as discussed in Chapters 2 and 3. As a result, we may see an even larger relative change in the density.



Figure 5.1. Calculations of relative changes in carrier densities achievable in TiN through electrical gating for varying film thicknesses. Close to 15 % relative change is expected in case of monolayer TiN

Gate dependent Hall measurements on a 1 nm TiN film were performed to test the feasibility of electrical control. The fabricated device consisted of four gold arms connected to the ends of the Hall cross, which served as the electrodes. A 35 nm thick alumina layer grown using atomic layer deposition (ALD) was used as the top gate oxide. A fifth gold arm is used for the top gate electrode. The schematic of the gate dependent Hall measurements is shown in Figure 5.2b. We observe close to 8.5% change in the carrier density in 1 nm TiN films, as plotted in Figure 5.2a.



Figure 5.2. (a) Experimentally retrieved carrier density as a function of bias gate voltage in 1-nm-thick TiN film with close to 8.5% change in carrier density; (b) schematic of the gate dependent Hall measurements. A 35-nm-thick layer of Al_2O_3 is used as the gate oxide and gold as the back-gate electrode.

5.3 Phase Modulation with Atomically Thin TiN

A dynamically tunable phase modulator was designed based on the tunability of atomically thin TiN. The phase modulation metasurface consists of an array of Au strips with a width of 258 nm and a 50 nm thickness placed on top of a 1 nm thick TiN film separated by a 45 nm Al_2O_3 spacer from a Au underlayer, which supports gap plasmon modes as schematically shown in Fig. 5.3a. To design our dynamic phase modulator, the carrier density in 1 nm thick TiN films was reduced by 12%, a number that is achievable with existing gating approaches. The calculated reflection coefficients and phases of the optimized structure with the TiN carrier densities reduced by 6% and 12% along with the unperturbed TiN film (0% change in carrier densities) are shown in Fig. 5.3b. As the TiN carrier density decreases, the reflection dip displays a slight blueshift as shown in Fig. 5.3b. The near zero reflection coefficient and 180° phase shift near or at 1.540 μ m marked by 6% in Figs. 5.3b represent the critical coupling state of the optimized structure. The reflection coefficients with the unperturbed and 12% lower carrier density are both larger than the 6% case at resonant wavelength, which demonstrates the breaking of balance between the resonant scattering channel and non-resonant channel. By increasing the TiN carrier density, the critical coupling state of the Au strip-TiN film hybrid metasurface moves into an over-coupling state, and the phase evolves from 180° to 0°, and ultimately to -180° at 1.552 μ m. With increasing TiN carrier density (red curve), the metasurface covering a 140° phase shift at 1.526 μ m moves into the under-coupling state. Therefore, along with the realization of different coupling states by gating ultrathin TiN films, the reflection phases are dynamically modulated at certain wavelengths.



Figure 5.3. (a) Schematic of Au strip-TiN film hybrid metasurface. (b) Change in reflection phase and coefficient at 1.550 μ m as a function of the change in charge carrier density. A phase modulation of >300° is achieved by lower the charge carrier density from 6% lower to 9%

5.4 Summary

The electrical tunability of the charge carrier concentration in atomically thin TiN has been demonstrated. Based on the theoretically achievable changes in the charge carrier density, the designed dynamic metasurface design enables >300° phase modulation at 1.55 μ m, which is achieved by using realistic numbers for carrier density changes in 1 nm thick TiN film. Despite the significant progress made in optical and electrical modulation, the development of active metasurfaces in the visible and near-infrared wavelength ranges is still greatly pursued, as these are spectral regions of utmost importance for sensing and optical signal processing technologies. Ultra-thin TiN film, an emerging electrically tunable material, offers an alternative material platform to realize dynamic light manipulation in the visible and NIR wavelength ranges.

6. Conclusion and Outlook

This thesis describes the development and characterization of plasmonic atomically thin TiN for tunable nanophotonic devices. We have experimentally demonstrated transdimensional TiN films down to thicknesses of 1 nm that retain their plasmonic behavior in the visible and near-infrared wavelength range. Moreover, the thickness dependent optical properties of TD TiN have been experimentally characterized using spectroscopy ellipsometry and theoretically investigated using both density functional theory calculations and a nonlocal Drude response model. The strong vertical confinement in TD TiN results in a consistent red shift of the optical properties and a decrease in the Drude plasma frequency in thinner films. In addition to the number of monolayers, surface oxidation and substrate properties also play a crucial role in controlling the properties of the atomically thin TiN. Most importantly, we have experimentally verified the nonlocal Drude response model, demonstrating the confinement induced thickness dependent plasma frequency in TD TiN.

Transdimensional materials can achieve remarkably large tunability of their response via free-carrier-assisted modulation due to their lower free carrier concentration compared to bulk metals and conventional thin films. Careful consideration of the thickness-dependent optical properties is required to optimize the modulation strength and device performance. Additionally, thinner films would allow control over the optical properties by adjusting the strain and lattice mismatch in the layer, facilitating a new way to engineer the optical response of the ultrathin films. While the tunable response of TCOs and 2D materials have been widely explored, studies of the TDMs' optical behavior have been limited.

Preliminary studies on TDMs show that they could exhibit strongly confined, tunable plasmons in the NIR and visible wavelengths, leading to a variety unique light-matter interactions. For example, it has been theorized that these tightly confined plasmons may enable radiative transitions in atomic-scale emitters that were considered inaccessible. Enabling access to these forbidden atomic transitions may significantly advance several different applications, including spectroscopy techniques, sensors based on forbidden transitions, as well as tunable ultrafast light generation in the near- and mid-infrared. Additionally, confined plasmons in ultra-thin structures allow for a large concentration of electromagnetic energy which could lead to strong coupling with nearby quantum emitters. Furthermore, studies have shown TiN to have strong nonlinear properties with an increased damage threshold compared to other metal films. Due to the strong confinement in observed in the atomically thin TiN, ultrathin films are expected to show an enhanced nonlinear response, which would be beneficial for all-optical devices such as ultrafast switches and frequency converters.

The realization of tunable nanophotonic devices with TDMs requires a comprehensive understanding of their plasmonic response. Appropriate models need to be developed to properly describe the evolution of the optical behavior and to design TDM-based tunable flat optical devices, or metasurfaces. Additionally, the synthesis of smooth, epitaxial, ultrathin films is critical in probing novel size effects in TDMs. As numerical modeling and material growth techniques for atomically thin materials mature, the realization of TDMbased tunable structures is expected to unlock new optical phenomena and photonic device concepts. Along with 2D materials, TDMs hold a great promise for realizing a new generation of tunable metasurfaces with quantum and nonlocal effects particularly in visible and NIR wavelength ranges.

REFERENCES

[1] S. A. Maier, *Plasmonics: Fundamentals and Applications*, en. Springer Science & Business Media, May 16, 2007, Google-Books-ID: yT2ux7TmDc8C, ISBN: 978-0-387-37825-1.

[2] W. A. Murray and W. L. Barnes, "Plasmonic materials," en, *Advanced Materials*, vol. 19, no. 22, pp. 3771–3782, 2007, ISSN: 1521-4095. DOI: 10.1002/adma.200700678.

[3] M. I. Stockman, K. Kneipp, S. I. Bozhevolnyi, S. Saha, A. Dutta, J. Ndukaife, N. Kinsey, H. Reddy, U. Guler, V. M. Shalaev, A. Boltasseva, B. Gholipour, H. N. S. Krishnamoorthy, K. F. MacDonald, C. Soci, N. I. Zheludev, V. Savinov, R. Singh, P. G. s, C. Lienau, M. Vadai, M. L. Solomon, D. R. Barton, M. Lawrence, J. A. Dionne, S. V. Boriskina, R. Esteban, J. Aizpurua, X. Zhang, S. Yang, D. Wang, W. Wang, T. W. Odom, N. Accanto, d. P. M. Roque, I. M. Hancu, L. Piatkowski, v. N. F. Hulst, and M. F. Kling, "Roadmap on plasmonics," en, *Journal of Optics*, vol. 20, no. 4, p. 043 001, Mar. 2018, ISSN: 2040-8986. DOI: 10.1088/2040-8986/aaa114.

[4] H. A. Atwater and A. Polman, "Plasmonics for improved photovoltaic devices," en, *Nature Materials*, vol. 9, no. 3, pp. 205–213, Mar. 2010, ISSN: 1476-1122. DOI: 10.1038/nmat2629.

[5] W. Li, U. Guler, N. Kinsey, G. V. Naik, A. Boltasseva, J. Guan, V. M. Shalaev, and A. V. Kildishev, "Refractory plasmonics with titanium nitride: Broadband metamaterial absorber," en, *Advanced Materials*, vol. 26, no. 47, pp. 7959–7965, Dec. 1, 2014, ISSN: 1521-4095. DOI: 10.1002/adma.201401874.

[6] D. K. Gramotnev and S. I. Bozhevolnyi, "Plasmonics beyond the diffraction limit," En, *Nature Photonics*, vol. 4, no. 2, p. 83, Feb. 2010, ISSN: 1749-4893. DOI: 10.1038/nphoton. 2009.282.

[7] S. Saha, A. Dutta, N. Kinsey, A. V. Kildishev, V. M. Shalaev, and A. Boltasseva, "On-chip hybrid photonic-plasmonic waveguides with ultrathin titanium nitride films," *ACS Photonics*, vol. 5, no. 11, pp. 4423–4431, Nov. 21, 2018. DOI: 10.1021/acsphotonics.8b00885.

[8] C. Haffner, D. Chelladurai, Y. Fedoryshyn, A. Josten, B. Baeuerle, W. Heni, T. Watanabe, T. Cui, B. Cheng, S. Saha, D. L. Elder, L. R. Dalton, A. Boltasseva, V. M. Shalaev, N. Kinsey, and J. Leuthold, "Low-loss plasmon-assisted electro-optic modulator," en, *Nature*, vol. 556, no. 7702, pp. 483–486, Apr. 2018, ISSN: 1476-4687. DOI: 10.1038/s41586-018-0031-4.

[9] S. Kawata, Y. Inouye, and P. Verma, "Plasmonics for near-field nano-imaging and superlensing," en, *Nature Photonics*, vol. 3, no. 7, pp. 388–394, Jul. 2009, ISSN: 1749-4893. DOI: 10.1038/nphoton.2009.111.

[10] K. A. Willets, A. J. Wilson, V. Sundaresan, and P. B. Joshi, "Super-resolution imaging and plasmonics," *Chemical Reviews*, vol. 117, no. 11, pp. 7538–7582, Jun. 14, 2017, ISSN: 0009-2665. DOI: 10.1021/acs.chemrev.6b00547.

[11] N. Kinsey, M. Ferrera, V. M. Shalaev, and A. Boltasseva, "Examining nanophotonics for integrated hybrid systems: A review of plasmonic interconnects and modulators using traditional and alternative materials [invited]," EN, *JOSA B*, vol. 32, no. 1, pp. 121–142, Jan. 1, 2015, ISSN: 1520-8540. DOI: 10.1364/JOSAB.32.000121.

[12] N. Kinsey, C. DeVault, J. Kim, M. Ferrera, V. M. Shalaev, Boltasseva, and A, "Epsilonnear-zero al-doped zno for ultrafast switching at telecom wavelengths," en, *Optica*, vol. 2, no. 7, pp. 616–622, ISSN: 2334-2536. DOI: 10.1364/OPTICA.2.000616.

[13] Y. Yang, K. Kelley, E. Sachet, S. Campione, T. S. Luk, J.-P. Maria, M. B. Sinclair, and I. Brener, "Femtosecond optical polarization switching using a cadmium oxide-based perfect absorber," *Nature Photonics*, vol. 11, no. 6, pp. 390–395, ISSN: 1749-4885. DOI: 10.1038/nphoton.2017.64.

[14] J. N. Anker, W. P. Hall, O. Lyandres, N. C. Shah, J. Zhao, and R. P. Van Duyne, "Biosensing with plasmonic nanosensors," en, *Nature Materials*, vol. 7, no. 6, pp. 442–453, Jun. 2008, ISSN: 1476-1122. DOI: 10.1038/nmat2162.

[15] D. Rodrigo, O. Limaj, D. Janner, D. Etezadi, d. F. J. G. Abajo, V. Pruneri, and H. Altug, "Mid-infrared plasmonic biosensing with graphene," en, *Science*, vol. 349, no. 6244, pp. 165–168, Jul. 10, 2015, PMID: 26160941, ISSN: 0036-8075, 1095-9203. DOI: 10.1126/ science.aab2051.

[16] M. Mesch, B. Metzger, M. Hentschel, and H. Giessen, "Nonlinear plasmonic sensing," Nano Letters, vol. 16, no. 5, pp. 3155–3159, May 11, 2016, ISSN: 1530-6984. DOI: 10.1021/acs.nanolett.6b00478.

[17] M. Liu, X. Yin, E. Ulin-Avila, B. Geng, T. Zentgraf, L. Ju, F. Wang, and X. Zhang, "A graphene-based broadband optical modulator," *Nature*, vol. 474, no. 7349, pp. 64–67, ISSN: 0028-0836. DOI: 10.1038/nature10067.

[18] J. Dionne, K. Diest, L. Sweatlock, and H. Atwater, "Plasmostor: A metal— oxide— si field effect plasmonic modulator," en, *Nano Letters*, vol. 9, no. 2, p. 897, ISSN: 1530-6984, 1530-6992. DOI: 10.1021/nl803868k.

[19] G. V. Naik, V. M. Shalaev, and A. Boltasseva, "Alternative plasmonic materials: Beyond gold and silver," en, *Advanced Materials*, vol. 25, no. 24, pp. 3264–3294, Jun. 25, 2013, ISSN: 1521-4095. DOI: 10.1002/adma.201205076.

[20] R. Chapman and P. Mulvaney, "Electro-optical shifts in silver nanoparticle films," *Chemical Physics Letters*, vol. 349, no. 5, pp. 358–362, Dec. 7, 2001, ISSN: 0009-2614. DOI: 10.1016/S0009-2614(01)01145-9.

[21] C. Novo and P. Mulvaney, "Charge-induced rayleigh instabilities in small gold rods," Nano Letters, vol. 7, no. 2, pp. 520–524, Feb. 1, 2007, ISSN: 1530-6984. DOI: 10.1021/nl062649t.

[22] Y.-W. Huang, H. W. H. Lee, R. Sokhoyan, R. A. Pala, K. Thyagarajan, S. Han, D. P. Tsai, and H. A. Atwater, "Gate-tunable conducting oxide metasurfaces," *Nano Letters*, vol. 16, no. 9, pp. 5319–5325, ISSN: 1530-6984. DOI: 10.1021/acs.nanolett.6b00555.

[23] G. V. Naik, J. Kim, and A. Boltasseva, "Oxides and nitrides as alternative plasmonic materials in the optical range [invited]," EN, *Optical Materials Express*, vol. 1, no. 6, pp. 1090–1099, Oct. 1, 2011, ISSN: 2159-3930. DOI: 10.1364/OME.1.001090.

[24] M. A. Noginov, L. Gu, J. Livenere, G. Zhu, A. K. Pradhan, R. Mundle, M. Bahoura, Y. A. Barnakov, and V. A. Podolskiy, "Transparent conductive oxides: Plasmonic materials for telecom wavelengths," *Applied Physics Letters*, vol. 99, no. 2, pp. 2014–2017, ISSN: 00036951. DOI: 10.1063/1.3604792.

[25] J. Kim, G. V. Naik, A. V. Gavrilenko, K. Dondapati, V. I. Gavrilenko, S. M. Prokes, O. J. Glembocki, V. M. Shalaev, and A. Boltasseva, "Optical properties of gallium-doped zinc oxide—a low-loss plasmonic material: First-principles theory and experiment," *Physical Review X*, vol. 3, no. 4, p. 041037, Dec. 31, 2013. DOI: 10.1103/PhysRevX.3.041037.

[26] N. K. Emani, V. A. Kildishev, V. M. Shalaev, and A. Boltasseva, "Graphene: A dynamic platform for electrical control of plasmonic resonance," *Nanophotonics*, vol. 4, no. 1, ISSN: 2192-8606. DOI: 10.1515/nanoph-2015-0014.

[27] N. Dabidian, I. Kholmanov, A. B. Khanikaev, K. Tatar, S. Trendafilov, S. H. Mousavi, C. Magnuson, R. S. Ruoff, and G. Shvets, "Electrical switching of infrared light using graphene integration with plasmonic fano resonant metasurfaces," *ACS Photonics*, vol. 2, no. 2, pp. 216–227, Feb. 18, 2015. DOI: 10.1021/ph5003279.

[28] A. M. Shaltout, V. M. Shalaev, and M. L. Brongersma, "Spatiotemporal light control with active metasurfaces," *Science*, vol. 364, no. 6442, ISSN: 10959203. DOI: 10.1126/science. aat3100.

[29] A. K. Geim and K. S. Novoselov, "The rise of graphene," en, *Nature Materials*, vol. 6, no. 3, pp. 183–191, Mar. 2007, ISSN: 1476-4660. DOI: 10.1038/nmat1849.

[30] N. Rivera, I. Kaminer, B. Zhen, J. D. Joannopoulos, and M. Soljačić, "Shrinking light to allow forbidden transitions on the atomic scale," en, *Science*, vol. 353, no. 6296, pp. 263–269, Jul. 15, 2016, PMID: 27418505, ISSN: 0036-8075, 1095-9203. DOI: 10.1126/science.aaf6308.

[31] F. J. García de Abajo, "Graphene plasmonics: Challenges and opportunities," ACS Photonics, vol. 1, no. 3, pp. 135–152, Mar. 19, 2014. DOI: 10.1021/ph400147y.

[32] A. Boltasseva and V. M. Shalaev, "Transdimensional photonics," *ACS Photonics*, vol. 6, no. 1, pp. 1–3, Jan. 16, 2019. DOI: 10.1021/acsphotonics.8b01570.

[33] A. Manjavacas and d. F. J. G. Abajo, "Tunable plasmons in atomically thin gold nanodisks," en, *Nature Communications*, vol. 5, p. 3548, Mar. 27, 2014, ISSN: 2041-1723. DOI: 10.1038/ncomms4548.

[34] H. Qian, Y. Xiao, and Z. Liu, "Giant kerr response of ultrathin gold films from quantum size effect," en, *Nature Communications*, vol. 7, p. 13153, Oct. 10, 2016, ISSN: 2041-1723. DOI: 10.1038/ncomms13153.

[35] S. Laref, J. Cao, A. Asaduzzaman, K. Runge, P. Deymier, R. W. Ziolkowski, M. Miyawaki, and K. Muralidharan, "Size-dependent permittivity and intrinsic optical anisotropy of nanometric gold thin films: A density functional theory study," en, *Optics Express*, vol. 21, no. 10, p. 11827, May 20, 2013, ISSN: 1094-4087. DOI: 10.1364/OE.21.011827.

[36] S. D. Campbell, R. W. Ziolkowski, J. Cao, S. Laref, K. Muralidharan, and P. Deymier, "Anisotropic permittivity of ultra-thin crystalline au films: Impacts on the plasmonic response of metasurfaces," *Applied Physics Letters*, vol. 103, no. 9, p. 091106, Aug. 26, 2013, ISSN: 0003-6951, 1077-3118. DOI: 10.1063/1.4819770.

[37] W. Ming, S. Blair, and F. Liu, "Quantum size effect on dielectric function of ultrathin metal film: A first-principles study of al(1\hspace0.167em1\hspace0.167em1)," en, *Journal of Physics: Condensed Matter*, vol. 26, no. 50, p. 505 302, Nov. 2014, ISSN: 0953-8984. DOI: 10.1088/0953-8984/26/50/505302.

[38] I.-B. Lin, T. W.-H. Sheu, and J.-H. Li, "Effects of exchange correlation functional on optical permittivity of gold and electromagnetic responses," EN, *Optics Express*, vol. 22, no. 25, pp. 30725–30734, Dec. 15, 2014, ISSN: 1094-4087. DOI: 10.1364/OE.22.030725.

[39] H. Qian, Y. Xiao, D. Lepage, L. Chen, and Z. Liu, "Quantum electrostatic model for optical properties of nanoscale gold films," *Nanophotonics*, vol. 4, no. 1, pp. 413–418, 2015, ISSN: 2192-8606. DOI: 10.1515/nanoph-2015-0022.

[40] I. V. Bondarev and V. M. Shalaev, "Universal features of the optical properties of ultrathin plasmonic films," EN, *Optical Materials Express*, vol. 7, no. 10, pp. 3731–3740, Oct. 1, 2017, ISSN: 2159-3930. DOI: 10.1364/OME.7.003731.

[41] D. Shah, A. Catellani, H. Reddy, N. Kinsey, V. Shalaev, A. Boltasseva, and A. Calzolari, "Controlling the plasmonic properties of ultrathin tin films at the atomic level," *ACS Photonics*, vol. 5, no. 7, pp. 2816–2824, Jul. 18, 2018. DOI: 10.1021/acsphotonics.7b01553.

[42] R. Sundararaman, T. Christensen, Y. Ping, N. Rivera, J. D. Joannopoulos, M. Soljačić, and P. Narang, "Plasmonics in argentene," en, Jun. 7, 2018, [Online; accessed 2019-10-18]. [Online]. Available: https://arxiv.org/abs/1806.02672v1.

[43] A. R. Echarri, J. D. Cox, and d. F. J. G. Abajo, "Quantum effects in the acoustic plasmons of atomically thin heterostructures," EN, *Optica*, vol. 6, no. 5, pp. 630–641, May 20, 2019, ISSN: 2334-2536. DOI: 10.1364/OPTICA.6.000630.

[44] I. V. Bondarev, H. Mousavi, and V. M. Shalaev, "Optical response of finite-thickness ultrathin plasmonic films," en, *MRS Communications*, vol. 8, no. 3, pp. 1092–1097, Sep. 2018, publisher: Cambridge University Press, ISSN: 2159-6859, 2159-6867. DOI: 10.1557/mrc. 2018.153.

[45] I. V. Bondarev, H. Mousavi, and V. M. Shalaev, "Transdimensional epsilon-near-zero modes in planar plasmonic nanostructures," *Physical Review Research*, vol. 2, no. 1, p. 013 070, Jan. 23, 2020, publisher: American Physical Society. DOI: 10.1103/PhysRevResearch.2.013070.

[46] G. X. Ni, A. S. McLeod, Z. Sun, L. Wang, L. Xiong, K. W. Post, S. S. Sunku, B.-Y. Jiang, J. Hone, C. R. Dean, M. M. Fogler, and D. N. Basov, "Fundamental limits to graphene plasmonics," en, *Nature*, vol. 557, no. 7706, pp. 530–533, May 2018, ISSN: 1476-4687. DOI: 10.1038/s41586-018-0136-9.

[47] Z. M. Abd El-Fattah, V. Mkhitaryan, J. Brede, L. Fernández, C. Li, Q. Guo, A. Ghosh, A. R. Echarri, D. Naveh, F. Xia, J. E. Ortega, and F. J. García de Abajo, "Plasmonics in atomically thin crystalline silver films," *ACS Nano*, vol. 13, no. 7, pp. 7771–7779, Jul. 23, 2019, ISSN: 1936-0851. DOI: 10.1021/acsnano.9b01651.

[48] T. Nagao, S. Yaginuma, T. Inaoka, and T. Sakurai, "One-dimensional plasmon in an atomic-scale metal wire," *Phys. Rev. Lett.*, vol. 97, p. 116802, 11 Sep. 2006. DOI: 10.1103/PhysRevLett.97.116802. [Online]. Available: https://link.aps.org/doi/10.1103/PhysRevLett. 97.116802.

[49] F. Moresco, M. Rocca, T. Hildebrandt, and M. Henzler, "Plasmon confinement in ultrathin continuous ag films," *Phys. Rev. Lett.*, vol. 83, pp. 2238–2241, 11 Sep. 1999. DOI: 10.1103/PhysRevLett.83.2238. [Online]. Available: https://link.aps.org/doi/10.1103/PhysRevLett.83.2238.

[50] E. Rosencher, A. Fiore, B. Vinter, V. Berger, P. Bois, and J. Nagle, "Quantum engineering of optical nonlinearities," en, *Science*, vol. 271, no. 5246, pp. 168–173, Jan. 12, 1996, ISSN: 0036-8075, 1095-9203. DOI: 10.1126/science.271.5246.168.

[51] N. Kinsey, A. A. Syed, D. Courtwright, C. DeVault, C. E. Bonner, V. I. Gavrilenko, V. M. Shalaev, D. J. Hagan, E. W. Van Stryland, and A. Boltasseva, "Effective third-order nonlinearities in metallic refractory titanium nitride thin films," en, *Optical Materials Express*, vol. 5, no. 11, p. 2395, Nov. 1, 2015, ISSN: 2159-3930. DOI: 10.1364/OME.5.002395.

[52] H. Qian, S. Li, C.-F. Chen, S.-W. Hsu, S. E. Bopp, Q. Ma, A. R. Tao, and Z. Liu, "Large optical nonlinearity enabled by coupled metallic quantum wells," en, *Light: Science & Applications*, vol. 8, no. 1, p. 13, Jan. 23, 2019, number: 1 publisher: Nature Publishing Group, ISSN: 2047-7538. DOI: 10.1038/s41377-019-0123-4.

[53] A. Walsh, C. R. A. Catlow, K. H. L. Zhang, and R. G. Egdell, "Control of the bandgap states of metal oxides by the application of epitaxial strain: The case of indium oxide," *Physical Review B*, vol. 83, no. 16, p. 161 202, Apr. 12, 2011. DOI: 10.1103/PhysRevB.83. 161202.

[54] Y.-J. Lu, R. Sokhoyan, W.-H. Cheng, G. K. Shirmanesh, A. R. Davoyan, R. A. Pala, K. Thyagarajan, and H. A. Atwater, "Dynamically controlled purcell enhancement of visible spontaneous emission in a gated plasmonic heterostructure," en, *Nature Communications*, vol. 8, no. 1, p. 1631, Nov. 21, 2017, ISSN: 2041-1723. DOI: 10.1038/s41467-017-01870-0.

[55] M. C. Sherrott, P. W. C. Hon, K. T. Fountaine, J. C. Garcia, S. M. Ponti, V. W. Brar, L. A. Sweatlock, and H. A. Atwater, "Experimental demonstration of >230° phase modulation in gate-tunable graphene–gold reconfigurable mid-infrared metasurfaces," *Nano Letters*, vol. 17, no. 5, pp. 3027–3034, May 10, 2017, publisher: American Chemical Society, ISSN: 1530-6984. DOI: 10.1021/acs.nanolett.7b00359.

[56] Z. Fei, A. S. Rodin, G. O. Andreev, W. Bao, A. S. McLeod, M. Wagner, L. M. Zhang, Z. Zhao, M. Thiemens, G. Dominguez, M. M. Fogler, A. H. C. Neto, C. N. Lau, F. Keilmann, and D. N. Basov, "Gate-tuning of graphene plasmons revealed by infrared nano-imaging," en, *Nature*, vol. 487, no. 7405, pp. 82–85, Jul. 2012, number: 7405 publisher: Nature Publishing Group, ISSN: 1476-4687. DOI: 10.1038/nature11253.

[57] D. Shah, Z. A. Kudyshev, S. Saha, V. M. Shalaev, and A. Boltasseva, "Transdimensional material platforms for tunable metasurface design," en, *MRS Bulletin*, vol. 45, no. 3, pp. 188–195, Mar. 2020, publisher: Cambridge University Press, ISSN: 0883-7694, 1938-1425. DOI: 10.1557/mrs.2020.63.

[58] R. A. Maniyara, D. Rodrigo, R. Yu, J. Canet-Ferrer, D. S. Ghosh, R. Yongsunthon, D. E. Baker, A. Rezikyan, d. F. J. G. Abajo, and V. Pruneri, "Tunable plasmons in ultrathin metal films," en, *Nature Photonics*, vol. 13, no. 5, pp. 328–333, May 2019, ISSN: 1749-4893. DOI: 10.1038/s41566-019-0366-x.

[59] X. Liu, J.-H. Kang, H. Yuan, J. Park, S. J. Kim, Y. Cui, H. Y. Hwang, and M. L. Brongersma, "Electrical tuning of a quantum plasmonic resonance," en, *Nature Nanotechnology*, vol. 12, no. 9, pp. 866–870, Sep. 2017, number: 9 publisher: Nature Publishing Group, ISSN: 1748-3395. DOI: 10.1038/nnano.2017.103.

[60] A. Kossoy, V. Merk, D. Simakov, K. Leosson, S. Kéna-Cohen, and S. A. Maier, "Optical and structural properties of ultra-thin gold films," en, *Advanced Optical Materials*, vol. 3, no. 1, pp. 71–77, Jan. 1, 2015, ISSN: 2195-1071. DOI: 10.1002/adom.201400345.

[61] D. Shah, H. Reddy, N. Kinsey, V. M. Shalaev, and A. Boltasseva, "Optical properties of plasmonic ultrathin tin films," en, *Advanced Optical Materials*, vol. 5, no. 13, p. 1700065, Jul. 1, 2017, ISSN: 2195-1071. DOI: 10.1002/adom.201700065.

[62] H. Van Bui, A. Y. Kovalgin, and R. A. M. Wolters, "On the difference between optically and electrically determined resistivity of ultra-thin titanium nitride films," *Applied Surface Science*, SURFINT-SREN III, vol. 269, pp. 45–49, Mar. 15, 2013, ISSN: 0169-4332. DOI: 10.1016/j.apsusc.2012.09.074.

[63] I. Doron-Mor, Z. Barkay, N. Filip-Granit, A. Vaskevich, and I. Rubinstein, "Ultrathin gold island films on silanized glass. morphology and optical properties," *Chemistry of Materials*, vol. 16, no. 18, pp. 3476–3483, Sep. 1, 2004, ISSN: 0897-4756. DOI: 10.1021/cm049605a.

[64] K. Leosson, A. S. Ingason, B. Agnarsson, A. Kossoy, S. Olafsson, and M. C. Gather, "Ultra-thin gold films on transparent polymers," *Nanophotonics*, vol. 2, no. 1, pp. 3–11, 2012, ISSN: 2192-8614. DOI: 10.1515/nanoph-2012-0030.

[65] N. Formica, D. S. Ghosh, A. Carrilero, T. L. Chen, R. E. Simpson, and V. Pruneri, "Ultrastable and atomically smooth ultrathin silver films grown on a copper seed layer," *ACS Applied Materials & Interfaces*, vol. 5, no. 8, pp. 3048–3053, Apr. 24, 2013, ISSN: 1944-8244. DOI: 10.1021/am303147w.

[66] D. I. Yakubovsky, Y. V. Stebunov, R. V. Kirtaev, G. A. Ermolaev, M. S. Mironov, S. M. Novikov, A. V. Arsenin, and V. S. Volkov, "Ultrathin and ultrasmooth gold films on monolayer mos2," en, *Advanced Materials Interfaces*, vol. 6, no. 13, p. 1900196, 2019, ISSN: 2196-7350. DOI: 10.1002/admi.201900196. [67] R. Lemasters, C. Zhang, M. Manjare, W. Zhu, J. Song, S. Urazhdin, H. J. Lezec, A. Agrawal, and H. Harutyunyan, "Ultrathin wetting layer-free plasmonic gold films," *ACS Photonics*, Oct. 3, 2019. DOI: 10.1021/acsphotonics.9b00907. [Online]. Available: https://doi.org/10.1021/acsphotonics.9b00907.

[68] U. Guler, J. C. Ndukaife, G. V. Naik, A. G. A. Nnanna, A. V. Kildishev, V. M. Shalaev, and A. Boltasseva, "Local heating with lithographically fabricated plasmonic titanium nitride nanoparticles," *Nano Letters*, vol. 13, no. 12, pp. 6078–6083, Dec. 11, 2013, ISSN: 1530-6984. DOI: 10.1021/nl4033457.

[69] S. Ishii, R. P. Sugavaneshwar, and T. Nagao, "Titanium nitride nanoparticles as plasmonic solar heat transducers," *The Journal of Physical Chemistry C*, vol. 120, no. 4, pp. 2343–2348, Feb. 4, 2016, ISSN: 1932-7447. DOI: 10.1021/acs.jpcc.5b09604.

[70] N. Kinsey, M. Ferrera, G. V. Naik, V. E. Babicheva, V. M. Shalaev, and A. Boltasseva, "Experimental demonstration of titanium nitride plasmonic interconnects," EN, *Optics Express*, vol. 22, no. 10, pp. 12238–12247, May 19, 2014, ISSN: 1094-4087. DOI: 10.1364/OE. 22.012238.

[71] V. E. Babicheva, N. Kinsey, G. V. Naik, M. Ferrera, A. V. Lavrinenko, V. M. Shalaev, and A. Boltasseva, "Towards cmos-compatible nanophotonics: Ultra-compact modulators using alternative plasmonic materials," en, *Optics Express*, vol. 21, no. 22, p. 27326, Nov. 4, 2013, ISSN: 1094-4087. DOI: 10.1364/OE.21.027326.

[72] L. Gui, S. Bagheri, N. Strohfeldt, M. Hentschel, C. M. Zgrabik, B. Metzger, H. Linnenbank, E. L. Hu, and H. Giessen, "Nonlinear refractory plasmonics with titanium nitride nanoantennas," *Nano Letters*, vol. 16, no. 9, pp. 5708–5713, Sep. 14, 2016, ISSN: 1530-6984. DOI: 10.1021/acs.nanolett.6b02376.

[73] P. Patsalas, N. Kalfagiannis, and S. Kassavetis, "Optical properties and plasmonic performance of titanium nitride," en, *Materials*, vol. 8, no. 6, pp. 3128–3154, May 29, 2015. DOI: 10.3390/ma8063128.

[74] G. M. Matenoglou, C. E. Lekka, L. E. Koutsokeras, G. Karras, C. Kosmidis, G. A. Evangelakis, and P. Patsalas, "Structure and electronic properties of conducting, ternary tixta1-xn films," *Journal of Applied Physics*, vol. 105, no. 10, p. 103714, May 15, 2009, ISSN: 0021-8979. DOI: 10.1063/1.3131824.

[75] A. Ern V.; Switendick, "Electronic band structure of tic, tin and tio," *Phys. Rev.*, vol. 137, no. 6A, p. 1927, 1965.

[76] W. B. Pearson, *Structure Reports: TiN.* Oosthoek, Scheltema and Holkema for International Union of Crystallography, 1993, vol. 11.

[77] A. L. I. V. A. Gubanov and V. P. Zhukov, *Electronic Structure of Refractory Carbides and Nitrides*. Cambridge University Press, Cambridge, 1994.

[78] P. Patsalas, C. Charitidis, and S. Logothetidis, "In situ and real-time ellipsometry monitoring of submicron titanium nitride titanium silicide electronic devices electronic structure of refractory carbides and nitrides," *Appl. Surf. Sci.*, vol. 154-155, pp. 256–262, 2000.

[79] E. Langereis, S. Heil, H. Knoops, W. Keuning, M. van de Sanden, and W. Kessels, "In situ spectroscopic ellipsometry as a versatile tool for studying atomic layer deposition," *J. Phys. D: Appl. Phys.*, vol. 42, 073001(1–19), 2009.

[80] H. Höchst, R. D. Bringans, P. Steiner, and T. Wolf, "Photoemission study of the electronic structure of stoichiometric and substoichiometric tin and zrn," *Phys. Rev. B*, vol. 25, pp. 7183–7191, 1982.

[81] C. G. H. Walker, C. A. Anderson, A. McKinley, N. Brown, and A. Joyce, "A combined aes, resonant photoemission and eels study of in-situ grown titanium nitride," *Surf. Sci.*, vol. 383, pp. 254–260, 1997.

[82] J. Pflüger, J. Fink, W. Weber, K. P. Bohnen, and G. Crecelius, "Dielectric properties of tic_x, tin_x, vc_x , and vn_x from 1.5 to 40 ev determined by electron-energy-loss spectroscopy," *Physical Review B*, vol. 30, no. 3, pp. 1155–1163, Aug. 1, 1984. DOI: 10.1103/PhysRevB.30. 1155.

[83] A. A. Herzing, U. Guler, X. Zhou, A. Boltasseva, V. Shalaev, and T. B. Norris, "Electron energy loss spectroscopy of plasmon resonances in titanium nitride thin films," *Appl. Phys. Lett.*, vol. 108, p. 171107, 2016.

[84] D. Gall, I. Petrov, and J. E. Greene, "Epitaxial $sc1xti_xn(001)$: Optical and electronic transport properties," *J. Appl. Phys.*, vol. 89, pp. 401–409, 2001.

[85] W. J. Chou, G. Yu, and J. Huan, "Mechanical properties of tin thin film coatings on 304 stainless steel substrates," *Surf. Coat. Technol.*, vol. 149, pp. 7–13, 2002.

[86] S. Fayeulle and P. Ehni, "Friction and wear behavior of tin in air: The chemistry of transfer films and debris formation," *Wear*, vol. 149, pp. 375–394, 1991.

[87] E. R. Margine, A. N. Kolmogorov, M. Reese, M. Mrovec, C. Elsässer, B. Meyer, R. Drautz, and P. D. G., "Development of orthogonal tight-binding models for ti-c and ti-n systems," *Phys. Rev. B*, vol. 84, p. 155 120, 2011.

[88] A. Delin, O. Eriksson, R. Ahuja, B. Johansson, M. S. S. Brooks, T. Gasche, S. Auluck, and J. Wills, "Optical properties of the group-ivb refractory metal compounds," *Phys. Rev. B*, vol. 54, p. 1673, 1996.

[89] A. Catellani and A. Calzolari, "Plasmonic properties of refractory titanium nitride," *Physical Review B*, vol. 95, no. 11, p. 115145, Mar. 27, 2017. DOI: 10.1103/PhysRevB.95. 115145.

[90] F. Ingason A.S.and Magnus, J. Agustsson, S. Olafsson, and J. Gudmundsson, "In-situ electrical characterization of ultrathin tin films grown by reactive dc magnetron sputtering on sio2," *Thin Sol. Films*, vol. 517, pp. 6731–6736, 2009.

[91] R. Kroger, M. Eizenberg, C. Marcadal, and L. Chen, "Plasma induced microstructural, compositional, and resistivity changes in ultrathin chemical vapor deposited titanium nitride films," *J. Appl. Phys.*, vol. 91, p. 5149, 2002.

[92] J. S. Chawla, X. Y. Zhang, and D. Gall, "Effective electron mean free path in tin(001)," *Journal of Applied Physics*, vol. 113, no. 6, p. 063704, Feb. 11, 2013, ISSN: 0021-8979. DOI: 10.1063/1.4790136.

[93] H. Van Bui, A. Kovalgin, J. Schmitz, and R. A. M. Wolters, "Conduction and electric field effect in ultra-thin tin films," *Appl. Phys. Lett.*, vol. 103, p. 051 904, 2013.

[94] G. Hoffmann, R. Berndt, and P. Johansson, "Two-electron photon emission from metallic quantum wells," *Physical Review Letters*, vol. 90, no. 4, p. 046803, Jan. 28, 2003. DOI: 10.1103/PhysRevLett.90.046803.

[95] M. König, S. Wiedmann, C. Brüne, A. Roth, H. Buhmann, L. W. Molenkamp, X.-L. Qi, and S.-C. Zhang, "Quantum spin hall insulator state in hgte quantum wells," en, *Science*, vol. 318, no. 5851, pp. 766–770, Nov. 2, 2007, PMID: 17885096, ISSN: 0036-8075, 1095-9203. DOI: 10.1126/science.1148047.

[96] d. F. J. G. Abajo and A. Manjavacas, "Plasmonics in atomically thin materials," en, *Faraday Discussions*, vol. 178, no. 0, pp. 87–107, May 20, 2015, ISSN: 1364-5498. DOI: 10. 1039/C4FD00216D.

[97] B. Frank, A. Ruff, W. Sigle, L. Fu, P. Kahl, S. Ludwigs, v. P. Aken, W. Osten, z. F.-J. M. Heringdorf, P. V. Braun, P. Mulvaney, T. Weiss, and H. W. Giessen, "Electrochemical route to large-area mono-crystalline gold platelets for high-quality plasmonic applications," EN, [Online; accessed 2016-12-13], Bragg Gratings, Photosensitivity, and Poling in Glass Waveg-uides, Optical Society of America, Jul. 27, 2014, JTu3A.60. DOI: 10.1364/BGPP.2014. JTu3A.60. [Online]. Available: http://www.osapublishing.org/abstract.cfm?uri=BGPP-2014-JTu3A.60.

[98] G. V. Naik, J. L. Schroeder, X. Ni, A. V. Kildishev, T. D. Sands, and A. Boltasseva, "Titanium nitride as a plasmonic material for visible and near-infrared wavelengths," en, *Optical Materials Express*, vol. 2, no. 4, p. 478, Apr. 1, 2012, ISSN: 2159-3930. DOI: 10.1364/OME.2.000478.

[99] G. V. Naik, B. Saha, J. Liu, S. M. Saber, E. A. Stach, J. M. K. Irudayaraj, T. D. Sands, V. M. Shalaev, and A. Boltasseva, "Epitaxial superlattices with titanium nitride as a plasmonic component for optical hyperbolic metamaterials," en, *Proceedings of the National Academy of Sciences*, vol. 111, no. 21, pp. 7546–7551, May 27, 2014, PMID: 24821762, ISSN: 0027-8424, 1091-6490. DOI: 10.1073/pnas.1319446111.

- [100] H. S. Lim, S. B. Kang, I. S. Jeon, G. H. Choi, Y. W. Park, S. I. Lee, and J. T. Moon, "Atomic layer deposition- and chemical vapor deposition-tin top electrode optimization for the reliability of ta2o5 and al2o3 metal insulator silicon capacitor for 0.13 µm technology and beyond," en, *Japanese Journal of Applied Physics*, vol. 40, no. 4S, p. 2669, Apr. 2001, ISSN: 1347-4065. DOI: 10.1143/JJAP.40.2669.
- [101] J. Kim, H. Hong, S. Ghosh, K.-Y. Oh, and C. Lee, "Physical properties of highly conformal tin thin films grown by atomic layer deposition," en, *Japanese Journal of Applied Physics*, vol. 42, no. 3R, p. 1375, Mar. 1, 2003, ISSN: 1347-4065. DOI: 10.1143/JJAP.42.1375.
- [102] D.-K. Lee, J.-J. Lee, and J. Joo, "Preparation of tin films at room temperature by inductively coupled plasma assisted chemical vapor deposition," English, *Surface and Coatings Technology*, vol. 171, no. 1-3, pp. 24–28, 2003, ISSN: 0257-8972. DOI: 10.1016/S0257-8972(03)00230-5.
- [103] D. Look, *Electrical Characterization of GaAs Materials and Devices*. New York, NY: Wiley, Jan. 1, 1989. [Online]. Available: http://corescholar.libraries.wright.edu/books/32.
- [104] U. Guler, S. Suslov, A. V. Kildishev, A. Boltasseva, and V. M. Shalaev, "Colloidal plasmonic titanium nitride nanoparticles: Properties and applications," *Nanophotonics*, vol. 4, no. 3, pp. 269–276, 2015, ISSN: 2192-8606. DOI: 10.1515/nanoph-2015-0017.
- [105] C. M. Zgrabik and E. L. Hu, "Optimization of sputtered titanium nitride as a tunable metal for plasmonic applications," EN, *Optical Materials Express*, vol. 5, no. 12, pp. 2786– 2797, Dec. 1, 2015, ISSN: 2159-3930. DOI: 10.1364/OME.5.002786.
- [106] V. P. Drachev, U. K. Chettiar, A. V. Kildishev, H.-K. Yuan, W. Cai, and V. M. Shalaev, "The ag dielectric function in plasmonic metamaterials," EN, *Optics Express*, vol. 16, no. 2, pp. 1186–1195, Jan. 21, 2008, ISSN: 1094-4087. DOI: 10.1364/OE.16.001186.
- [107] S. M. Rossnagel and T. S. Kuan, "Alteration of cu conductivity in the size effect regime," Journal of Vacuum Science & Technology B: Microelectronics and Nanometer Structures Processing, Measurement, and Phenomena, vol. 22, no. 1, pp. 240–247, Jan. 1, 2004, ISSN: 1071-1023. DOI: 10.1116/1.1642639.

- [108] A. Hafaiedh and N. Bouarissa, "Quantum confinement effects on energy gaps and electron and hole effective masses of quantum well aln," *Physica E: Low-dimensional Systems and Nanostructures*, vol. 43, no. 9, pp. 1638–1641, Jul. 2011, ISSN: 1386-9477. DOI: 10.1016/j.physe.2011.05.013.
- [109] F. Long, W. E. Hagston, P. Harrison, and T. Stirner, "The structural dependence of the effective mass and luttinger parameters in semiconductor quantum wells," *Journal of Applied Physics*, vol. 82, no. 7, pp. 3414–3421, Oct. 1, 1997, ISSN: 0021-8979, 1089-7550. DOI: 10.1063/1.365657.
- [110] J. P. Perdew, K. Burke, and M. Ernzerhof, "Generalized gradient approximation made simple," *Physical Review Letters*, vol. 77, no. 18, pp. 3865–3868, Oct. 28, 1996. DOI: 10.1103/ PhysRevLett.77.3865.
- [111] N. Troullier and J. L. Martins, "Efficient pseudopotentials for plane-wave calculations," *Physical Review B*, vol. 43, no. 3, pp. 1993–2006, Jan. 15, 1991. DOI: 10.1103/PhysRevB.43. 1993.
- [112] P. Giannozzi, S. Baroni, N. Bonini, M. Calandra, R. Car, C. Cavazzoni, D. Ceresoli, G. L. Chiarotti, M. Cococcioni, I. Dabo, A. D. Corso, d. S. Gironcoli, S. Fabris, G. Fratesi, R. Gebauer, U. Gerstmann, C. Gougoussis, A. Kokalj, M. Lazzeri, L. Martin-Samos, N. Marzari, F. Mauri, R. Mazzarello, S. Paolini, A. Pasquarello, L. Paulatto, C. Sbraccia, S. Scandolo, G. Sclauzero, A. P. Seitsonen, A. Smogunov, P. Umari, and R. M. Wentzcovitch, "Quantum espresso: A modular and open-source software project for quantum simulations of materials," en, *Journal of Physics: Condensed Matter*, vol. 21, no. 39, p. 395 502, 2009, ISSN: 0953-8984. DOI: 10.1088/0953-8984/21/39/395502.
- [113] A. Calzolari, A. Ruini, and A. Catellani, "Transparent conductive oxides as near-ir plasmonic materials: The case of al-doped zno derivatives," *ACS Photonics*, vol. 1, no. 8, pp. 703–709, Aug. 20, 2014. DOI: 10.1021/ph500118y.
- [114] M. Eaton, A. Catellani, and A. Calzolari, "Vo2 as a natural optical metamaterial," Opt. Express, vol. 26, pp. 5342–5357,
- [115] S. Campione, I. Brener, and F. Marquier, "Theory of epsilon-near-zero modes in ultrathin films," *Physical Review B*, vol. 91, no. 12, p. 121408, Mar. 16, 2015. DOI: 10.1103/ PhysRevB.91.121408.
- [116] S. Vassant, J.-P. Hugonin, F. Marquier, and J.-J. Greffet, "Berreman mode and epsilon near zero mode," EN, *Optics Express*, vol. 20, no. 21, pp. 23971–23977, Oct. 8, 2012, ISSN: 1094-4087. DOI: 10.1364/OE.20.023971.

- [117] I. Milošv, H.-H. Strehblow, B. Navinšek, and M. Metikoš-Huković, "Electrochemical and thermal oxidation of tin coatings studied by xps," en, *Surface and Interface Analysis*, vol. 23, no. 7-8, pp. 529–539, Jul. 1, 1995, ISSN: 1096-9918. DOI: 10.1002/sia.740230713.
- [118] N. Jiang, H. J. Zhang, S. N. Bao, Y. G. Shen, and Z. F. Zhou, "Xps study for reactively sputtered titanium nitride thin films deposited under different substrate bias," *Physica B: Condensed Matter*, vol. 352, no. 1, pp. 118–126, Oct. 30, 2004, ISSN: 0921-4526. DOI: 10. 1016/j.physb.2004.07.001.
- [119] Y. Cheng and Y. F. Zheng, "Surface characterization and mechanical property of tin/ticoated niti alloy by piiid," *Surface and Coatings Technology*, The Eighth International Workshop on Plasma-Based Ion Implantation and Deposition, vol. 201, no. 15, pp. 6869–6873, Apr. 23, 2007, ISSN: 0257-8972. DOI: 10.1016/j.surfcoat.2006.09.055.
- [120] L. Braic, N. Vasilantonakis, A. Mihai, I. J. Villar Garcia, S. Fearn, B. Zou, N. M. Alford, B. Doiron, R. F. Oulton, S. A. Maier, A. V. Zayats, and P. K. Petrov, "Titanium oxynitride thin films with tunable double epsilon-near-zero behavior for nanophotonic applications," *ACS Applied Materials & Interfaces*, vol. 9, no. 35, pp. 29857–29862, Sep. 6, 2017, ISSN: 1944-8244. DOI: 10.1021/acsami.7b07660.
- [121] F. Magnus, A. S. Ingason, S. Olafsson, and J. T. Gudmundsson, "Growth and in-situ electrical characterization of ultrathin epitaxial tin films on mgo," *Thin Solid Films*, vol. 519, no. 18, pp. 5861–5867, Jul. 1, 2011, ISSN: 0040-6090. DOI: 10.1016/j.tsf.2011.02.078.
- [122] D. A. King and D. P. Woodruff, Growth and Properties of Ultrathin Epitaxial Layers. English. Burlington: Elsevier, 1997, OCLC: 476128654, ISBN: 978-0-08-053267-7. [Online]. Available: http://www.123library.org/book_details/?id=34819.
- [123] A. Rodríguez Echarri, E. J. H. Skjølstrup, T. G. Pedersen, and F. J. García de Abajo, "Theory of electron energy-loss spectroscopy in atomically thin metallic films," *Physical Review Research*, vol. 2, no. 2, p. 023096, Apr. 28, 2020, publisher: American Physical Society. DOI: 10.1103/PhysRevResearch.2.023096.
- [124] Y. Silberberg and T. Sands, "Optical properties of metallic quantum wells," *IEEE Journal of Quantum Electronics*, vol. 28, no. 7, pp. 1663–1669, Jul. 1992, event: IEEE Journal of Quantum Electronics, ISSN: 1558-1713. DOI: 10.1109/3.142553.
- [125] F. H. da Jornada, L. Xian, A. Rubio, and S. G. Louie, "Universal slow plasmons and giant field enhancement in atomically thin quasi-two-dimensional metals," en, *Nature Communications*, vol. 11, no. 1, p. 1013, Feb. 21, 2020, ISSN: 2041-1723. DOI: 10.1038/s41467-020-14826-8.

- [126] I. V. Bondarev and C. M. Adhikari, "Collective excitations and optical response of ultrathin carbon-nanotube films," *Physical Review Applied*, vol. 15, no. 3, p. 034 001, Mar. 1, 2021, publisher: American Physical Society. DOI: 10.1103/PhysRevApplied.15.034001.
- [127] L. Vertchenko, L. Leandro, E. Shkondin, O. Takayama, I. V. Bondarev, N. Akopian, and A. V. Lavrinenko, "Cryogenic characterization of titanium nitride thin films," EN, *Optical Materials Express*, vol. 9, no. 5, pp. 2117–2127, May 1, 2019, publisher: Optica Publishing Group, ISSN: 2159-3930. DOI: 10.1364/OME.9.002117.
- [128] M. Gioti, J. Arvanitidis, D. Christofilos, K. Chaudhuri, T. Zorba, G. Abadias, D. Gall, V. M. Shalaev, A. Boltasseva, and P. Patsalas, "Plasmonic and phononic properties of epitaxial conductive transition metal nitrides," en, *Journal of Optics*, vol. 22, no. 8, p. 084001, Jun. 2020, publisher: IOP Publishing, ISSN: 2040-8986. DOI: 10.1088/2040-8986/ab97c6.
- [129] W.-P. Guo, R. Mishra, C.-W. Cheng, B.-H. Wu, L.-J. Chen, M.-T. Lin, and S. Gwo, "Titanium nitride epitaxial films as a plasmonic material platform: Alternative to gold," *ACS Photonics*, vol. 6, no. 8, pp. 1848–1854, Aug. 21, 2019, publisher: American Chemical Society. DOI: 10.1021/acsphotonics.9b00617.
- [130] H. Reddy, U. Guler, Z. Kudyshev, A. V. Kildishev, V. M. Shalaev, and A. Boltasseva, "Temperature-dependent optical properties of plasmonic titanium nitride thin films," ACS Photonics, vol. 4, no. 6, pp. 1413–1420, Jun. 21, 2017. DOI: 10.1021/acsphotonics.7b00127.
- [131] J. A. Briggs, G. V. Naik, Y. Zhao, T. A. Petach, K. Sahasrabuddhe, D. Goldhaber-Gordon, N. A. Melosh, and J. A. Dionne, "Temperature-dependent optical properties of titanium nitride," en, *Applied Physics Letters*, vol. 110, no. 10, p. 101901, Mar. 6, 2017, publisher: AIP Publishing LLC AIP Publishing, ISSN: 0003-6951. DOI: 10.1063/1.4977840.
- [132] A. S. Roberts, M. Chirumamilla, D. Wang, L. An, K. Pedersen, N. A. Mortensen, and S. I. Bozhevolnyi, "Ultra-thin titanium nitride films for refractory spectral selectivity [invited]," EN, *Optical Materials Express*, vol. 8, no. 12, pp. 3717–3728, Dec. 1, 2018, ISSN: 2159-3930. DOI: 10.1364/OME.8.003717.
- [133] R. Sato, S. Ishii, T. Nagao, M. Naito, and Y. Takeda, "Broadband plasmon resonance enhanced third-order optical nonlinearity in refractory titanium nitride nanostructures," ACS *Photonics*, vol. 5, no. 9, pp. 3452–3458, Sep. 19, 2018, publisher: American Chemical Society. DOI: 10.1021/acsphotonics.8b00357.
- [134] A. Catellani and A. Calzolari, "Tailoring the plasmonic properties of ultrathin tin films at metal-dielectric interfaces," EN, *Optical Materials Express*, vol. 9, no. 3, pp. 1459–1468, Mar. 1, 2019, ISSN: 2159-3930. DOI: 10.1364/OME.9.001459.

- [135] A. Catellani, P. D'Amico, and A. Calzolari, "Tailoring the plasmonic properties of metals: The case of substoichiometric titanium nitride," *Physical Review Materials*, vol. 4, no. 1, p. 015 201, Jan. 27, 2020, publisher: American Physical Society. DOI: 10.1103 / PhysRevMaterials.4.015201.
- [136] W. A. Britton, Y. Chen, and L. D. Negro, "Double-plasmon broadband response of engineered titanium silicon oxynitride," EN, *Optical Materials Express*, vol. 9, no. 2, pp. 878– 891, Feb. 1, 2019, publisher: Optica Publishing Group, ISSN: 2159-3930. DOI: 10.1364/OME. 9.000878.
- [137] B. Saha, S. Saber, G. V. Naik, A. Boltasseva, E. A. Stach, E. P. Kvam, and T. D. Sands, "Development of epitaxial al_xsc1xn for artificially structured metal/semiconductor superlattice metamaterials," *physica status solidi* (b), vol. 252, no. 2, pp. 251–259, 2015, _eprint : *https* : //onlinelibrary.wiley.com/doi/pdf/10.1002/pssb.201451314, ISSN: 1521-3951. DOI: 10.1002/pssb.201451314.
- [138] L. V. Keldysh, "Coulomb interaction in thin semiconductor and semimetal films," Soviet Journal of Experimental and Theoretical Physics Letters, vol. 29, p. 658, Jun. 1, 1979, ADS Bibcode: 1979JETPL..29..658K, ISSN: 0021-3640.
- [139] N. S. Rytova, "Screened potential of a point charge in a thin film," Moscow University Physics Bulletin, vol. 3, no. 30, 1967, arXiv: 1806.00976. [Online]. Available: http://arxiv. org/abs/1806.00976.
- [140] I. V. Bondarev, "Finite-thickness effects in plasmonic films with periodic cylindrical anisotropy [invited]," EN, *Optical Materials Express*, vol. 9, no. 1, pp. 285–294, Jan. 1, 2019, publisher: Optica Publishing Group, ISSN: 2159-3930. DOI: 10.1364/OME.9.000285.
- [141] C. M. Adhikari and I. V. Bondarev, "Controlled exciton-plasmon coupling in a mixture of ultrathin periodically aligned single-wall carbon nanotube arrays," en, *Journal of Applied Physics*, vol. 129, no. 1, p. 015 301, Jan. 4, 2021, publisher: AIP Publishing LLCAIP Publishing, ISSN: 0021-8979. DOI: 10.1063/5.0031212.
- [142] I. V. Bondarev, O. L. Berman, R. Y. Kezerashvili, and Y. E. Lozovik, "Crystal phases of charged interlayer excitons in van der waals heterostructures," en, *Communications Physics*, vol. 4, no. 1, pp. 1–13, Jun. 25, 2021, ISSN: 2399-3650. DOI: 10.1038/s42005-021-00624-1.
- [143] Z. Sun, J. Beaumariage, Q. Wan, H. Alnatah, N. Hougland, J. Chisholm, Q. Cao, K. Watanabe, T. Taniguchi, B. M. Hunt, I. V. Bondarev, and D. Snoke, "Charged bosons made of fermions in bilayer structures with strong metallic screening," *Nano Letters*, vol. 21, no. 18, pp. 7669–7675, Sep. 22, 2021, publisher: American Chemical Society, ISSN: 1530-6984. DOI: 10.1021/acs.nanolett.1c02422.

- [144] P. Cudazzo, I. V. Tokatly, and A. Rubio, "Dielectric screening in two-dimensional insulators: Implications for excitonic and impurity states in graphane," *Physical Review B*, vol. 84, no. 8, p. 085 406, Aug. 19, 2011, publisher: American Physical Society. DOI: 10.1103/ PhysRevB.84.085406.
- [145] R. H. Ritchie, "Plasma losses by fast electrons in thin films," *Physical Review*, vol. 106, no. 5, pp. 874–881, Jun. 1, 1957, publisher: American Physical Society. DOI: 10.1103/PhysRev.106.874.
- [146] S. Raza, G. Toscano, A.-P. Jauho, M. Wubs, and N. A. Mortensen, "Unusual resonances in nanoplasmonic structures due to nonlocal response," *Physical Review B*, vol. 84, no. 12, p. 121 412, Sep. 29, 2011, publisher: American Physical Society. DOI: 10.1103/PhysRevB.84. 121412.
- [147] N. Saveskul, N. Titova, E. Baeva, A. Semenov, A. Lubenchenko, S. Saha, H. Reddy, S. Bogdanov, E. Marinero, V. Shalaev, A. Boltasseva, V. Khrapai, A. Kardakova, and G. Goltsman, "Superconductivity behavior in epitaxial tin films points to surface magnetic disorder," *Physical Review Applied*, vol. 12, no. 5, p. 054 001, Nov. 1, 2019, publisher: American Physical Society. DOI: 10.1103/PhysRevApplied.12.054001.
- [148] E. Sondheimer, "The mean free path of electrons in metals," Advances in Physics, vol. 1, no. 1, pp. 1–42, Jan. 1, 1952, publisher: Taylor & Francis, ISSN: 0001-8732. DOI: 10.1080/00018735200101151.

VITA

Deesha Shah received her Bachelor's in Science in Materials Science and Engineering from University of Illinois, Urbana-Champaign, Illinois in 2014, where she worked in the research group of John Rogers developing mechanically tunable plasmonic devices. In 2015, she joined the research group of Professors Alexandra Boltasseva and Vladimir Shalaev as a graduate research assistant in the School of Electrical and Computer Engineering and Birck Nanotechnology Center at Purdue University, West Lafayette, Indiana. Her research includes the development of plasmonic thin films, particularly transition metal nitrides, and noble metals. She specifically has focused on studying the optical properties of transdimensional metallic films as they approach thicknesses of a few monolayers and investigating their uses in tunable nanophotonic devices.

PUBLICATIONS

JOURNAL PUBLICATIONS

- D. Shah, M. Yang, Z. Kudyshev, V. M. Shaleav, I. V. Bondarev, A. Boltasseva, "Thickness dependent Drude plasma frequency in transdimensional plasmonic TiN," manuscript under review, (2022).
- D. Zhang, D. Shah, A. Boltasseva, Y. Gogotsi, "MXenes for Photonics," emphACS Photonics, 2022.
- S. Saha, D. Shah, V. M. Shalaev, A. Boltasseva, "Tunable Metasurfaces: Controlling Light in Space and Time," emphOptics and Photonics News 32 (7), 2021.
- A. H. Chu, B. Beauchamp, D. Shah, A. Dutta, A. Boltasseva, V. M. Shalaev, E. E. Marinero, "Hybrid Magneto Photonic Material Structure for Plasmon Assisted Magnetic Switching," *Opt. Mat. Express* 10 (12), 2020.
- 5. **D. Shah**, Z. Kudyshev, S. Saha, V. M. Shalaev, A. Boltasseva, "Transdimensional material platforms for tunable metasurface design," emphMRS Bulletin, 2020.
- C. C. Chiang, S. I Bogdanov, O. A Makarova, X. Xu, S. Saha, D. Shah, D. Wang, A. S Lagutchev, A.V Kildishev, A. Boltasseva, V. M Shalaev, "Chip-compatible quantum plasmonic launcher", *Adv. Optical Mater.* 8 (20), 2019.
- S. I. Bogdanov, O. A. Makarova, X. Xu, Z. O. Martin, A. S. Lagutchev, M. Olinde,
 D. Shah, S. N. Chowdhury, A. R. Gabidullin, I. A. Ryzhikov, I. A. Rodionov, A. V. Kildishev, S. I. Bozhevolnyi, Alexandra Boltasseva, V. M. Shalaev, J. B. Khurgin,
 "Ultrafast quantum photonics enabled by coupling plasmonic nanocavities to strongly radiative antennas", *Optica* 7 (5), 2020.
- H. Jiang, H. Reddy, D. Shah, Z. A. Kudyshev, S. Choudhury, D. Wang, Y. Jiang,
 A. V. Kildishev, "Modulating phase by metasurfaces with gated ultra-thin TiN films", Nanoscale 11 (23), 2019.

- D. Shah, A. Catellani, H. Reddy, N. Kinsey, V. M. Shalaev, A. Boltasseva, and A. Calzolari, "Controlling the Plasmonic Properties of Ultrathin TiN Films at the Atomic Level," ACS Photonics 5 (7), 2018.
- K. Chaudhuri, A. Shaltout, D. Shah, U. Guler, A. Dutta, V. M. Shalaev, A. Boltasseva, "Photonic spin Hall effect in robust phase gradient metasurfaces utilizing transition metal nitrides", ACS Photonics 6 (1), 2018.
- S. I. Bogdanov, M. Y. Shalaginov, A. S. Lagutchev, C. C. Chiang, D. Shah, A. S. Baburin, I. A. Ryzhikov, I. A. Rodionov, A. V. Kildishev, A. Boltasseva, V. M. Shalaev, "Ultrabright room-temperature sub-nanosecond emission from single nitrogen-vacancy centers coupled to nanopatch antennas", *Nanoletters* (18) 8, 2018.
- D. Shah, H. Reddy, N. Kinsey, V. M. Shalaev, and A. Boltasseva, "Optical Properties of Plasmonic Ultrathin TiN Films," Adv. Optical Mater. 5 (13), 2017.

CONFERENCES PROCEEDINGS

- D. Shah, M. Yang, Z. Kudyshev, V. M. Shaleav, I. V. Bondarev, A. Boltasseva, "Effect of electron confinement on the optical properties in transdimensional plasmonic TiN", Bulletin of the American Physical Society, APS March Meeting, 2022.
- D. Shah, V. M. Shalaev, A. Boltasseva, "Evolution of the optical properties in atomically thin plasmonic titanium nitride," Proc. SPIE 11682, Optical Components and Materials XVIII, 1168212, 2021.
- D. Shah, K. Chaudhuri, Z. Wang, A. Catellani, M. Alhabeb, H. Reddy, X. Meng, S. Azzam, N. Kinsey, A. Kildishev, Y. Kim, V. Shalaev, A. Calzolari, Y. Gogotsi, and A. Boltasseva, "New materials and approaches for tailorable nanophotonics (Conference Presentation)", Proc. SPIE 10922, Smart Photonic and Optoelectronic Integrated Circuits XXI, 109220V, 2019.
- 4. **D. Shah**, A. Catellani, H. Reddy, N. Kinsey, V. M. Shalaev, A. Calzolari, and A. Boltasseva "Ultra-thin transition metal nitrides for tailorable plasmonic devices (Con-

ference Presentation)", Proc. SPIE 10722, Plasmonics: Design, Materials, Fabrication; Characterization, and Applications XVI, 1072203, 2018.

- H. Reddy, D. Shah, N. Kinsey, V. M. Shalaev, A. Boltasseva, "Ultra-thin plasmonic metal nitrides: Tailoring optical properties to photonic applications," 2017 International Conference on Optical MEMS and Nanophotonics (OMN), 2017.
- D. Shah, H. Reddy, N. Kinsey, V. M. Shalaev and A. Boltasseva, "Optical properties of ultrathin plasmonic TiN films," 2017 Conference on Lasers and Electro-Optics (CLEO), 2017.